Teachers' Experiences of Teaching Creative Thinking Skills in Lesotho Classrooms

Ву

'MARETS'EPILE MAUREEN MOLETE

200202785

A Dissertation Submitted in Partial Fulfillment of the Requirements

for the Award of the Degree of Masters of Education in the

Department of Educational Foundations

Faculty of Education

National University of Lesotho

July 2023

CERTIFICATION

This is to certify that this dissertation has been read and approved as having met the requirements of the Faculty of Education, the National University of Lesotho for the award

of Masters in Education (M.ED)	
Supervisor (Dr Julia Chere-Masopha)	
Head of Department	
Dean of Faculty of Education	

COPYRIGHT STATEMENT

According to the university's intellectual property policy, the National University of Lesotho has the copyright to this research report. Without previous explicit consent from the University, no part of the work may be duplicated, saved in a retrieval system, or transmitted in any way, including through analogue and digital media. However, under the rules and restrictions of the Lesotho Copyright laws, excerpts or extracts from this dissertation may be copied with full attribution to the author and the University.

DECLARATION

I hereby state that, except for places where other sources have been cited, this dissertation, Teachers' Experiences of Teaching Creative Thinking Skills in Lesotho Classrooms, is entirely my own work. I further attest that I have never before submitted this dissertation for a degree at any other university.

'Marets'epile Molete

July 2023

ACKNOWLEDGEMENTS

Firstly, I would like to thank God for letting me through all these difficulties. I have experienced your guidance day by day and I felt your presence in me, I thank you, Lord!

I sincerely acknowledge and express my heartfelt gratitude to my supervisor**Dr Julia Chere-Masopha** who made this work possible. Her guidance and advice carried me through all the stages of writing my dissertation. She was just not a supervisor to me but a real mother who could support me along this academic journey and I considered myself blessed to be supervised byher. From the bottom of my heart, I thank you!

I would like to thank my colleague Mrs. Mamakhoathi Teke for her brilliant comments and suggestions, thanks to you. It would not be easy to complete this study without the help of these people.

I would also like to give special thanks to my husband Seabata Molete for allowing me to pursue my studies. I further express my sincere gratitude to my sister Tankiso Motjope who assisted and supported me during my studies, and my children Retsepile Molete (son) and Makoi Molete (daughter) for the love they showed me during this hard time, for their continuous support and understanding when undertaken my research. Your prayers for me sustained me this far.

I wish to acknowledge my sister-in-law Mamoroesi Linda Molete for taking care of my children during this difficult time. You showed me that you are my family, and our love for each other will always be there forever.

Finally, I must express my sincere appreciation to my principal and deputy principal for allowing me to further my studies and to interact with teachers, as well as all the participating teachers who contributed to information that made this study a success.

DEDICATION

I am dedicating this study to the following beloved people who have meant and continue to mean so much to me. First and foremost, to my late mother (Matankiso Motjope) who believed in me and taught me to trust and believe in hard work. From her, I learned that there is nothing greater than word of God and with Him anything is possible.

Next, I remember my mother-in-law (Mangaka Molete) for her words of love and guidance, things that only mothers say. Still, I pray that you can see that you have been my mother and always forever shall you be!

To my children, Retsepile (son) and Makoi (daughter), I always appreciated your wise and encouraging words. You highly cherished the route I took to further my studies and even if I could not fulfill your needs, you never gave up on me. I love you!

TABLE OF CONTENTS

CERTIFICAT	TION	.ii
COPYRIGH	T STATEMENT	iii
DECLARAT	ION	iv
ACKNOWL	EDGEMENTS	.V
DEDICATIO)N	vi
TABLE OF (CONTENTSv	⁄ii
LIST OF TA	BLESx	(ii
LIST OF AB	BREVIATIONS AND ACRONYMSxi	iii
ABSTRACT	xi	iv
CHAPTER 1	: BACKGROUND TO THE STUDY1	.5
1.1	INTRODUCTION	.5
1.2	BACKGROUND TO THE STUDY	.5
1.2.1	Defining Creative Thinking Skills	.5
1.2.2	Benefits of Teaching Creative Thinking Skills	.6
1.2.3	Teachers' Practices of Teaching Creative Thinking Skills in the Classroom	.7
1.2.4	Challenges that Teachers Experience in Teaching Creative Thinking Skills 1	.8
1.3	PROBLEM STATEMENT	.8
1.4	RESEARCH PURPOSE AND QUESTIONS	9

	1.4.1	Research Aim and Objectives	19
	1.5	SIGNIFICANCE OF THE STUDY	20
	1.6	THEORETICAL FRAMEWORK	20
	1.7	RESEARCH METHODOLOGY	20
	1.7.1	Epistemology: Interpretivism	20
	1.7.2	Paradigm: Constructivism	21
	1.7.3	Research Approach: Qualitative	21
	1.7.4	Research Methods and instruments: Semi-structured Interviews	21
	1.8	DEFINITION OF KEY CONCEPTS	21
	1.9	SUMMARY	22
	1.10	DISSERTATION LAYOUT STRUCTURE	23
С	HAPTER 2	: RELATED REVIEW OF LITERATURE	24
	2.1	INTRODUCTION	24
	2.2	THEORETICAL FRAMEWORK: CONSTRUCTIVISM	24
	2.2.1	Teaching and Learning Strategies Associated with Constructivism	24
	2.3	CONCEPTUAL FRAMEWORK: CREATIVE THINKING	25
	2.3.1	Definition of Creative Thinking Skills	25
	2.4	RESEARCH ON CREATIVE THINKING SKILLS	26
	2 / 1	The Benefits of Teaching Creative Thinking Skills	26

2.4.2	Teachers' Practices of Teaching Creative Thinking Skills	. 27
2.4.3	Factors Leading to the Challenges that Teachers Experience in Teaching Creat	tive
	Thinking Skills	. 27
2.4.4	Strategies for Developing Creative Thinking Skills	. 29
2.5	STUDIES ON TEACHING CREATIVE THINKING SKILLS	. 30
2.6	SUMMARY	. 31
CHAPTER 3	: RESEARCH METHODOLOGY	. 32
3.1	INTRODUCTION	. 32
3.2	METHODOLOGY	. 32
3.2.1	Research Epistemology: Interpretivism	. 32
3.2.2	Research Paradigm: Constructivism	. 33
3.2.3	Research Approach: Qualitative	. 33
3.2.4	Research Design: Case Study	. 34
3.2.5	Population and Participants' Selection	. 34
3.2.6	Research Methods and Instruments	. 35
3.2.7	Ethical Considerations and Trustworthiness	. 38
3.3	SUMMARY	. 40
CHAPTER 4	: FINDINGS	. 41
<i>1</i> 1	INTRODUCTION	<i>1</i> 1

4.2	THE PARTICIPANTS' PROFILE	41
4.3	PRESENTATION OF THE RESULTS	42
4.3.1	Teachers' Interpretation of Creative Thinking Skills	43
4.3.2	Teachers' Practices of Teaching Creative Thinking Skills	46
4.3.3	Factors Leading to the Challenges that Teachers Experience in Teaching CTS	48
4.3.5	Teachers Views on how they are supported to Teach CTS	52
4.3.6	Teachers' General Views about Teaching CTS in Lesotho classrooms	53
4.4	SUMMARY	56
CHAPTER 5	: DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS	57
5.1	INTRODUCTION	57
5.2	MAJOR FINDINGS	57
5.2.1	Teachers' Interpretation of CTS	57
5.2.2	Teachers' Views on the Benefits of CTS in the Classroom	58
5.2.3	Teachers' Practices of Teaching CTS in the Classroom	59
5.2.4	Factors Leading to the Challenges that Teachers Experience in Teaching CTS	60
5.2.4	How Teachers are Supported to Teach CTS	61
5.2.5	Teachers' General Views about Teaching CTS in Lesotho Classrooms	61
5.3	CONCLUSIONS	62
5.4	LIMITATIONS OF THE STUDY	63

	5.5	RECOMMENDATIONS	63
	5.6	SUMMARY	64
R	EFERENCE	ES	65
A	PPENDICE	<u>-</u> S	77
	APPEND	IX 1: THE LETTER TO THE PARTICIPANT	77
	APPEND	IX 2: PARTICIPANTS' STATEMENT OF CONSENT	78
	APPEND	IX 3:SEMI-STRUCTURED INTERVIEW PROTOCOL	79
	APPEND	IX 4: FACULTY LETTER	81
	APPEND	IX 5: THE LETTER FROM THE RESEARCHER TO THE PRINCIPAL	82
	APPEND	IX 6:GUIDING INTERVIEW QUESTIONS	83
	APPEND	IX7: LANGUAGE EDITOR'S LETTER	85
	APPEND	IX 8: TURNITIN CERTIFICATE	86

LIST OF TABLES

Table 4.1: Summary of participants' professional profile
--

LIST OF ABBREVIATIONS AND ACRONYMS

BED Bachelor's Degree in Education

CAP Curriculum Assessment Policy

CTS Creative Thinking Skills

DEPRG Directions Evidence for Educational Research Group

DES Diploma in Education Secondary

JC Junior Certificate

LCE Lesotho College of Education

LGSCE Lesotho General Certificate for Secondary Education

MOET Ministry of Education and Training

NTTC National Teacher Training College

NUL National University of Lesotho

OECD The Organization for Economic Cooperation and Development

PISA Programme for International Student Assessment

STC Secondary Teacher Certificate

UNESCO United Nations Educational Scientific and Cultural Organization

ABSTRACT

The study investigated teachers' experiences of teaching creative thinking skills in Lesotho classrooms. The enhancement of creative thinking skills (CTS) is required for lifelong learning and as significant skill in the 21st century in Lesotho schools. Using a qualitative approach, data for this study was collected from eight teachers using semi-structured interviews. These teachers were intentionally selected from a High School in the Leribe district. This study revealed that teachers in Lesotho rarely teach CTS in their classrooms because they have inadequate knowledge and skills relating to teaching this skill, schools lack resources to assist teachers, and the Ministry of Education and Training's lack of support in the schools'CTS teaching. Thus, much as Lesotho's education policy on curriculum and assessment recommends this skill to be taught in the classroom, the conditions that exist in the schools comprise the teaching of CTS; therefore, this study recommends that the Ministry of Education and Training find ways of supervising and monitoring the implementation of CTS in Lesotho classrooms. The study also recommends that the Ministry should offer training programmes to develop school management and teachers' pedagogical skills relating to CTS.

Keywords: teacher experience, creativity, skills, and creative thinking skills

CHAPTER 1: BACKGROUND TO THE STUDY

1.1 INTRODUCTION

In Lesotho, there had been concerns about school education which was teacher-centred and examination-driven (Raselimo & Mahao, 2015). Teaching did not emphasize the requirements and assessment of the competencies and skills outlined in the curricula documents. As a response to these concerns, the Ministry of Education and Training introduced the Curriculum and Assessment Policy which emphasized teaching, learning, and assessment that provide and assess learners' skills, attitudes, and values that enable them to overcome the challenges of the 21st Century (MoET, 2009). This document further demands that teachers use classroom strategies to develop learners' creative thinking skills (MoET, 2009). In this regard, teachers are expected to teach creative thinking skills in Lesotho classrooms. Hence, this study intended to investigate teachers' experiences of teaching creative thinking skills (CTS) in Lesotho classrooms. This chapter (Chapter 1) introduces this study, and it is structured as follows: Introduction (this section), Background to the study, Problem statement, Research questions, Research aim and objectives, Purpose of the study, Significance, Methodology of the study, Definition of terms, and Summary.

1.2 BACKGROUND TO THE STUDY

In today's global competition and innovation world, creative thinking skills are becoming more valuable for personal and professional success (Robinson, 2011). Shaheen (2010) points out that it is for this reason that learners should be equipped with creative thinking skills so that in the future, they can compete in the global markets, which their success is driven by innovation and economic competitiveness. Puccio and Cabra (2010) support Shaheen by emphasizing that a learner needs to reach their creative thinking skills potential to avoid unemployment and to be able to survive in an innovative economy as well as to overcome the challenges they encounter in their communities and workplace.

1.2.1 Defining Creative Thinking Skills

Gafour and Gafour (2020) describe creative thinking as a cluster of skills needed to produce original and valuable ideas. This cluster of skills is considered an essential ability that an individual

needs to learn to address the daily challenges in the workplace and society (The Partnership for 21st Century and Learning, 2016). This set of skills characterizes an individual's readiness to produce new ideas.

According to Mishra and Kereluik (2013), the demand for creative thinking skills has been around for centuries, and in the 21st century, it has been pronounced in many and various contexts (Robinson, 2011). Thus, since the 19th century, several educators have been pushing schools to foster creative thinking skills in teaching (Swayer, 2015). Subsequently, many national education systems have begun to include creative thinking skills in their curriculum (The Partnership for 21st Century and Learning, 2016).

Creative thinking skills are now acknowledged as part of formal education, which promotes lifelong learning (Lucas & Venckute, 2020). For example, the Programme for International Student Assessment (PISA) now uses the Creative Thinking Framework to measure whether school systems prepare learners for the twenty-first-century global knowledge economy (OECD, 2019). This PISA for Creative Thinking Framework is used in the OECD countries, from North and South America to Europe and Asia-Pacific (OECD, 2019).

In Europe, creative thinking skills are also recognized as vital for progress in providing knowledge to society and innovation-driven economies (OECD, 2019). In most sub-Saharan countries, such as Kenya, Mali, and Zambia, the curriculum emphasizes the learner's creative thinking skills development (UNESCO, 2020). Similarly, in Lesotho, the statement from The Curriculum and Assessment Policy (CAP) emphasizes that, by the time learners leave school, they should have developed creative thinking and entrepreneurial skills for productivity (Ministry of Education and Training, 2009).

1.2.2 Benefits of Teaching Creative Thinking Skills

Creativity has been viewed as the most important human attribute needed and wanted skill in the 21st Century (Akgunduz, Aydeniz, Cakmakci, Cavas, Colrlu, Oner, & Ozdemir, 2015; The Partnership for 21st Century skillsof Learning, 2016). Craig (2012) indicates that learners need creative thinking skills to make their way in a complex and constantly evolving future and to develop abilities to undertake work that machines cannot quickly reproduce. According to Khoiri (2019), creative thinking skills are an essential part of cognitive learning that aims to solve

problems encountered in the present or future. Lucas and Spencer (2017) view creative thinking skills as a necessary competence for today's young people to develop. Carson (2010) argues that learners with creative thinking skills can cope, adapt and flourish in their personal lives, education, businesses, and the society. As Lucas and Venckute (2020) described, creative thinking skills are significant in the 21st Century for promoting lifelong learning as part of formal education. As such, schools are expected to prepare learners to go beyond what they have been taught and learned in the classroom by demonstrating that they can think creatively (Schleicher, 2012).

Failure to develop learners' creative thinking skills can result in unemployable school graduates and experience extreme poverty (Jacobs & Tlali, 2015). This is why Hobbs, Ebrahimi, Cabral, Yoon and Al-Humaidan (2010) emphasize that in addition to classrooms, creative thinking skills should be taught at the workplace to equip workers with these skills to create ideas and become problem-solvers. These skills are crucial for employment in most businesses and organizations.

1.2.3 Teachers' Practices of Teaching Creative Thinking Skills in the Classroom

Teachers play an essential role in fostering learners' creative thinking abilities, as argued by the UNESCO International Bureau of Education (2014). They should focus on creative teaching methods that develop learners' creativity during the class (Alencar & Fleith, 2016). Teachers should encourage learners to practice self-evaluation and assessment of their ideas in a classroom context. These teachers should also improvise their teaching strategies to allow learners to construct and develop their knowledge (Besancon & Lubart, 2008; Li, 2016). According to Aripin, Sugandi, Mu'minah and Mulyani (2020), teachers should identify and use creative teaching strategies to develop learners' creative thinking skills and must create a learning process that accomplishes learners' thinking skills. Lucas and Spencer (2017) say these strategies should be practical, target application, and motivate learners to think creatively (OECD, 2019).

According to Craft (2011), teachers can utilise techniques in their classrooms to develop creative thinking skills, including establishing links between or among concepts, asking learners to reflect on the possibilities and solutions of one problem, and thinking critically about their ideas. The teachers should also develop learners' ability to generate creative ideas that improve how they predict the efficacy and consequences of their proposed solutions (Byrne, Shipman & Mumford,

2010). According to Zabelina and Robinson (2010), there are many creative thinking strategies that teachers can use to help learners to develop cognitive flexibility, which is the brain's ability to adapt to unplanned events, and they should also offer these learners opportunities to enhance the way they organize concepts and information.

1.2.4 Challenges that Teachers Experience in Teaching Creative Thinking Skills Despite the perceived benefits of creative thinking skills, it has been established that teachers do not teach these skills because of various reasons. For example, Tanujaya (2016) observes that for teachers to find strategies that effectively develop creative thinking is challenging. Tangaard (2011) also identifies time as one factor influencing teachers' reluctance to engage in creative teaching. Tangaard (2013) explained that teachers complain that the strategies used for teaching creative thinking require a lot of time and effort for preparation.

Consequently, the intensive preparation required for creative thinking lessons eats into the time of other subjects (Tangaard, 2013). Beghetto and Kaufman (2010) also established that the intensive planning for creative thinking skills puts much pressure on teachers to design learning tasks that foster learners' creative thinking, and yet at the same time, these teachers are expected to meet the requirements of other subjects they teach. However, even though the importance of developing learners' creative thinking skills has been well emphasized in the literature, there is still limited information on teachers' classroom practices with this strategy (Swayer, 2015). This has also been observed by Gafour and Gafour (2020), who observed that more information about creative skills would be helpful. There are very few studies that investigated how these skills should be taught. However, such studies are described in Chapter 2.

1.3 PROBLEM STATEMENT

The Curriculum and Assessment Policy (CAP) of 2009, emphasizes education that develops learners' creative skills and attitudes for achieving rapid social and economic change (MoET, 2009). In this policy, creative thinking is regarded as problem-solving oriented and capable of equipping learners with the skills that deal with the challenges they experience in everyday modern life (MoET, 2009). The curriculum also promotes creating, acquiring, and developing knowledge and skills (CAP, 2009), some of which are creative thinking skills. However, there are

no clear guidelines for how teachers should develop these skills. Studies focusing on teaching these skills since the policy introduced them could not be located. As it is, the information about how these skills are taught in Lesotho schools is scarce. Therefore, this study investigated teachers' experiences of teaching creative thinking skills in Lesotho classrooms.

1.4 RESEARCH PURPOSE AND QUESTIONS

This study investigated teachers experiences of teaching creative thinking skills in Lesotho classrooms. Thus, the main question asked in this study was: What are teachers experiences of teaching creative thinking skills in Lesotho classrooms? The questions that were used to collect datathat would answer the main question of this study were:

- 1. How do teachers in Lesotho classrooms interpret creative thinking skills?
- 2. What are teachers' practices of teaching creative thinking skills?
- 3. Which factors leading to the challenges teachers experience in teaching creative thinking skills?
- 4. How are teachers supported to teach creative thinking in the classroom?
- 5. What are teachers' general views about teaching creative thinking skills in Lesotho classrooms?

1.4.1 Research Aim and Objectives

This study aimed to collect information that explained teachers experiences in teaching creative thinking skills in Lesotho classrooms. Thus, the objectives of this study are:

- 1. To establish how teachers in Lesotho classrooms interpret creative thinking skills.
- 2. To investigate teachers practices in teaching creative thinking skills.
- 3. To identify the factors leading to the challenges that teachers experience in teaching creative thinking skills.
- 4. To establish how teachers are supported to teach creative thinking skills.
- 5. To establish teachers' general views of teaching creative thinking skills in Lesotho classrooms.

1.5 SIGNIFICANCE OF THE STUDY

This study attempted to shed light on teachers' experiences of teaching Creative Thinking Skills In Lesotho. The study will play a significant role in shring its findings with the Ministry of Education and NCDC, LCE and NUL about the status of teaching creative thinking skills in Lesotho classrooms. These bodies can use the information this study will generate to help teachers modify their teaching and learning from others who proposed creative thinking skills. The findings of this study will also help teachers find strategies that support them in developing creative thinking skills and emphasise the implementation of those teaching strategies that will enhance the development of creative thinking in Lesotho classrooms. The study will add value in how teachers in Lesotho schools can be supported to teach this skill in their classrooms and contribute to curriculum change.

1.6 THEORETICALFRAMEWORK

This study was guided by the theory of constructivism, which believes that learners use their existing experience to construct a reality that makes sense to them (Mascolo & Fisher, 2005). Winfree (2017) advocates that learning in constructivism theory involves learners developing their knowledge. Therefore, in this study, creative thinking skills teaching involves using constructivism strategies that encourage learners to develop practical skills to open up a platform where they have opportunities to create ideas and express themselves (Winfree, 2017).

1.7 RESEARCH METHODOLOGY

The methodology used in the study is structured in the following manner: Interpretivism epistemology, constructivism paradigm, and qualitative approach and research methods.

1.7.1 Epistemology: Interpretivism

The study used interpretivism epistemology, arguing that truth and knowledge are subjective based on people's experiences and their understanding of cultural and historical beliefs (Flick, 2014).

1.7.2 Paradigm: Constructivism

The study employed the constructivism paradigm, the mother of the interpretivism paradigm of philosophy. Their relationship is that the interpretivism perspective holds that individuals, in their reasoning, do not have access to the real world, suggesting that their knowledge of the perceived world is meaningful in its terms and can be understood through the careful use of Interpretivism procedures (Bereiter, 1994).

1.7.3 Research Approach: Qualitative

This study uses a qualitative research approach associated with the constructivist philosophical paradigm, which seeks to understand a phenomenon under study from the participants' experiences (Cresswell, 2013).

1.7.4 Research Methods and instruments: Semi-structured Interviews

In this study, the research method that was used was an interview. This study used semistructured interviews to collect participant data (Abawi, 2017).

1.8 DEFINITION OF KEY CONCEPTS

The key concepts in this study are **Teacher Experiences**, **Creativity**, **Skills and Creative thinkingskills**. The key concepts discussed in this section concern the teachers' experience of teaching CTS.

Teacher experience: In this study, the definition of teacher experience has been used in line with Harris and Sass (2011) definition. They describe teacher experience as teacher effectiveness influenced significantly by teacher personality factors, especially caring, intelligence, subject matter knowledge and teaching skills. In a Directions Evidence and Policy Research Group (DEPRG, 2014) report, teacher experience is defined as the number of years of teaching a teacher has in the classroom setting. In this study, the definition of Harris and Sass is adopted as it emphasizes the teacher effectiveness of teaching experience in the teaching and learning process.

Creativity: Schubert (2021) views creativity as the process that leads to an outcome that generates a positive effect. Further, Amabile and Pratt (2016) define creativity as a phenomenon whereby something new and valuable is formed. Even Scott (2015) defines creativity as the ability

to generate something new. Also, Apino and Retnawati (2017) state that creativity is the ability to generate new ideas. The study adopted the definitions of Amabile and Pratt (2016), Apino and Retnawati (2017) and Scott (2015), establishing that creativity involves the generation of something new to solve a problem.

Skills: According to Kassema (2016), skills is the ability of an individual to put into action the knowledge one has to accomplish the proposed goal. Gafour and Gafour (2020) view skills as the ability that requires one to produce original and valuable ideas with the expectation of a specific outcome. Both the definitions of Kassema (2016) and, Gafour and Gafour (2020) have been adopted as they have a similar meaning related to the context of this study.

Creative thinking skills: Craft (2011) describes creative skills as the ability to make or bring to existence something new, whether a new solution to a problem, a new method or a new object. Gardner (2008), Sternberg (2007) both cited in Scott (2015) define creative skills as the capacity to generate new ideas and solutions, break new ground, invoke fresh thinking, pose unfamiliar questions and arrive at unexpected answers. Gafour and Gafour (2020) view creative thinking skills as a set of abilities required to generate original and valuable ideas. Literature provides various definitions of creative skills, but this study adopts the definition of Sternberg (2001), cited in Scott (2015) and, Gafour and Gafour (2020), which emphasizes that learners need to acquire creative thinking skills in order to produce valuable ideas.

1.9 **SUMMARY**

This chapter intended to introduce the study investigating teachers' experiences of teaching thinking skills in Lesotho classrooms. In the background of this chapter, creative thinking skills are defined, benefits and the factors leading to the challenges that teachers experiencein teaching these skills have been discussed. In addition to the background, the study presented the purpose and the research questions of the study, the objectives and the significance of the study, the theoretical framework, the research methodology and the definition of key concepts, the summary (this section) and an outline of the study was introduced by in this chapter. The next Chapter will discuss the literature reviewed for this study.

1.10 DISSERTATION LAYOUT STRUCTURE

This study is organized into five chapters which are:

Chapter 1 provides an overview of this study. It outlines the study background to the problem statement, research purpose and questions, research aim and objectives, significance of the study, theoretical framework, research methodology, definition of terms, and Summary. Chapter 2 reports on the literature reviewed for this study this literature includes theoretical framework, conceptual framework and exiting studies on CTS. Chapter 3 describes the methodological approaches used for data collection and analysis. The chapter also explains how research ethics, trustworthiness, credibility and dependability were considered and ensured in this study. Chapter 4 presents the findings of this study. Chapter 5 summarizes and discusses the findings of this study, describes the study's limitations and makes recommendations.

CHAPTER 2: RELATED REVIEW OF LITERATURE

2.1 INTRODUCTION

This study aimed to investigate teachers' experience of teaching creative thinking skills in Lesotho classrooms. Chapter 1 introduced and gave an overview of this study. This chapter reports on the literature reviewed to inform this study. The chapter is structured into five main sections: Introduction (this section), Theoretical Framework, Conceptual Framework, Studies on Teaching Creative Thinking, and Summary. The report on the review of the studies carried out is presented under the empirical literature section.

2.2 THEORETICAL FRAMEWORK: CONSTRUCTIVISM

Many learning theories, including behaviourism and constructivism, guide the teaching and learning process. This study was guided by constructivism theory. Constructivism is a process in which new information is connected to previously held beliefs, resulting in the understanding and creation of new knowledge (Tomljenovic & Vorkapic, 2020). Piaget (1968) is the father of the constructivist view of learning. He believes a learner's interactions with the environment result in patterned and transformed knowledge. Apart from Piaget (1968), the other pioneer of constructivism is Vygotsky (1986), who believes that a teacher can intentionally nurture and teach learners in collaboration with them. Even Naylor and Keogh (1999) observe that learning in constructivism involves an active process in which learners construct meaning by linking their understanding and knowledge through experiences. Therefore, in this study, the teacher helps learners to construct or create their knowledge through experiences and apply creative thinking skills to overcome complex problems. Further, it contributes to widening learners thinking and reinforce their abilities in transforming ideas to creative applications.

2.2.1 Teaching and Learning Strategies Associated with Constructivism

In this study, the researcher employed constructivist teaching and learning strategies to develop creative thinking skills in the classroom. According to Gunduz and Hursen (2015), teachers that practice constructivist teaching strategies empower learners to access their previous experiences and beliefs, reshaping their prior knowledge in light of the applied class content. Karpov (2014) advocates that in a constructivist classroom, teachers enhance learners experiences to develop

their creative skills and must be encouraged to use innovative approaches to facilitate this development process. As observed by (Vygotsky, 2016), the activities such as brainstorming performed in a constructivist classroom lead to the development of learners' creative and critical abilities, which are then applied in the workplace and the community.

Therefore, the focus of this study is on constructivism theory since the constructivist teacher engages instructional strategies that direct the teaching and learning process that, is a learner-centered approach; thus, by following the constructivist approach of teaching in the classroom, the teacher can open up the scope of creative expression and think of the learners (Woolfolk, 2010). Winfree (2017) advocates that, in a constructivist classroom, learners develop their knowledge through thinking and understanding; hence they become creative learners who can make informed and thoughtful decisions.

2.3 CONCEPTUAL FRAMEWORK: CREATIVE THINKING

In recent years, and with rapid changes in globalization, creative thinking and creativity have become one of the most regarded skills to possess (Songkram, 2015). The view is mainly held by those interested in education. They argue that creative thinking skills should be fostered as one of the valued outcomes of schooling in the 21st century (The Partnership for 21st Century Skills of Learning, 2016; Griffin, McGaw & Care, 2012; Mishra & Kereluik, 2013; Adams, 2019). Zabelina and Robinson (2011) argue that learners with creative thinking skills tend to have high and flexible levels of cognitive control. In addition, creative thinking skills train learners to develop many ideas and arguments, ask questions, acknowledge the argument's truth, and make them open-minded and responsive to different perspectives (Arifin, 2017; Tendrita, Mohanal & Zubaidah, 2016). Even Grohman and Szmidt (2013) further point out that creative thinking can enhance teacher instruction and the learning process of the learners and is, therefore, an essential part of teacher training that can influence the professionalism of future teachers.

2.3.1 Definition of Creative Thinking Skills

Even though the importance of creative thinking is widely accepted, there is a lack of agreement regarding a definition of this concept among scholars. This results from the fact that creativity is a broad and difficult define concept, particularly in academic research; hence, defining creative thinking has similarly become difficult. Also, different scholars have defined this concept

differently (Reunamo, 2014). For example, Sternberg and Kaufman (2011), and Gafour and Gafour (2020) describe creative thinking skills as abilities required to generate unique and valuable ideas. Craft (2011) views this concept as an ability to make or bring to existence something new, whether a new solution to a problem, a new method or a new object. Similarly, Fredagsvik (2021) expounds on creative thinking as developing new ideas and finding new ways of looking at problems and opportunities. Daud, Omar, Turiman and Osman (2012) define creative thinking similarly to Craft (2011). They say it is a process in which a person can use thinking to develop new ideas in detail during the acquisition process. Scott (2015) supports both Craft (2011) and Daud et al. (2012) in their description of this concept, where they view it as the capacity to generate new ideas and solutions, break new ground, invoke fresh ways of thinking, pose unfamiliar questions and arrive at unexpected answers. Based on all these definitions, Hursen, Kaplan, and Ozdal (2014) emphasize that creative thinking ability should be acknowledged as essential to boosting innovation and as the main factor in developing one's competence. Similarly, Akgunduz et al. (2015) point out that creative thinking is now considered the most important human attribute needed in the 21st century.

Although these definitions do not vary that much in meaning, this study adopted the definition of Scott (2015), who describes creative thinking skills as the ability to generate something new. This definition can apply even in a teaching and learning situation where learners' skills are developed to use their knowledge to create new solutions for their learning problems or everyday life.

2.4 RESEARCH ON CREATIVE THINKING SKILLS

This section explored the related studies on the benefits of creative thinking skills (CTS), teachers practices of teaching CTS, the factors leading to the challenges that teachers experience in teaching CTS, and strategies for developing CTS. It also focused on studies reviewed on teaching CTS.

2.4.1 The Benefits of Teaching Creative Thinking Skills

Studies have revealed that teaching creative thinking is vital for developing learners' creative ability, which is considerable in this era (Albar & Southcott, 2021). Other scholars like Pesout and Nietfeld (2021) have also observed that creative thinking is considered a primary objective of

teaching and learning instruction that applies to today's market and employability. Accordingly, learning creative thinking skills has become an important goal in teacher instruction and is associated with promoting quality teaching (Geronik & Kimhi, 2019). This skill further allows learners to gain knowledge and experience to solve problems, and this has been affirmed by Hobbs et al. (2010), who stated that creative thinking skills help learners create ideas with their knowledge and become problem-solvers in the classroom. Creative thinking skills allow learners to gain knowledge and experiences and discover, recognize, and solve problems with multiple techniques (Zarkasyi, 2015).

2.4.2 Teachers' Practices of Teaching Creative Thinking Skills

In education, to develop learners' creativity, teachers have to create and assign learners learning activities capable of developing creative thinking skills (Ulfah, Prabawanto & Jupri, 2017). As observed by Selkrig and Keamy (2017), teachers should be informed or equipped with the skills that are appropriate for teaching creative skills and teaching for creativity.

Concerning how to teach creative thinking, the learning tasks developed and assigned by teachers should be divergent and progressive to allow learners to identify a problem, find solutions and communicate the results (Senel, 2019). This kind of task should provide learners with an opportunity to give more than one correct answer and encourage them to think differently and flexibly. Further, Lubart and Batton (2017) maintain that a lesson that encourages creativity must include divergent and convergent thinking. These learning activities should be designed to ensure that learners are at the forefront of learning activities or learner-centred (Sharma & Arif, 2015; Husaman, 2015). However, Sarwinda (2013) established that teachers should implement creative thinking skills that must be complemented by creative instructional planning.

2.4.3 Factors Leading to the Challenges that Teachers Experience in Teaching Creative Thinking Skills

Despite the value observed in creative thinking skills, it has been observed that studies conducted showed that there are factors leading to the challenges that teachers experience in teaching creative thinking skills in the classroom. However, learning institutions still use teaching strategies that do not promote this skill because of many challenges. (Al-Zoubei, 2014). Many studies associate such challenges with teachers. They observe that teachers do not possess the

skills needed to develop learners' creative thinking in the classroom (Laius and Rannikmae, 2011). Also other studies observed that teachers have poor self-confidence and a feeling of helplessness to practice teaching that promotes creative thinking skills (Suciawati, 2019; Al-Atoom et al. (2011). Teachers also find it challenging to attain strategies that effectively develop creative teaching (Tanujaya, 2016).

The time required for preparing for and teaching creative skills influences teachers' reluctance to engage in creative teaching. Teachers believe that the intensive planning required to design learning tasks that foster learners' creative thinking puts much pressure on them while, at the same time, they are expected to meet the requirements of other subjects they teach (Tangaard, 2013);

According to Chang (2013) teachers find it difficult to develop creative thinking skills if learners have limited knowledge and other thinking skills.

Teachers' classroom management and interaction with learners do not encourage creative thinking (Arvyati, Ibrahim, & Irawan, 2015).

Based on all these observations, Arvyati, Ibrahim, and Irawan (2015) suggest that teachers should be exposed to training programs that develop their abilities to identify and effectively use strategies that develop creative thinking skills in the classrooms. Suciawati (2019) also argues that a lack of self-confidence among learners towards their abilities, such as feeling embarrassed and afraid to show themselves and of being wrong, feeling tense working on questions, are some factors that complicate creative thinking teaching in the classroom.

Apart from teacher-related factors, Eragamreddy (2013) observe that schools also lack facilities for learners to develop the ability to think independently. They say that due to the lack of facilities, the learning process in schools emphasizes unproductive thinking, focuses on drilling learners to memorize, and looks for one correct answer to the questions. Yusnaeni, Corebima, Susilo, and Zubaidah (2017) observed that learners' unique creative thinking abilities require conditions that pay attention and involve each learner in the learning experiences that suit their abilities and learning styles.

2.4.4 Strategies for Developing Creative Thinking Skills

Juliantine (2019) argues that despite these factors and challenges, creative thinking must be developed in education to achieve high creative human resources in the workplace and society's everyday life. Even in human resource development, the emphasis is on promoting creative thinking ability in the knowledge of the economic era (Cheng-Shin Lin, 2016). UNESCO International Bureau of Education (2014) affirms that creative thinking skills should be fostered in all disciplines and intellectual and social areas. This is supported by Nuzliah (2015), who argues that it is often difficult to solve problems and create new ideas that overcome life and workplace challenges without creative thinking ability.

Thus, even where there are challenges in developing these skills, there should be a way of addressing these challenges. Some of the strategies that teachers may use to develop CTS is to design learning activities that are learner-centred and that ensure that learners are in the front position when these activities are executed (Sharma & Arif, 2015; Husaman, 2015). As observed by Azhari and Somakim(2014) selection of teaching methods that put learners at the center of learning and provide them with personal experiences will enhance the acquisition of CTS. This is supported by Jeng, Hsu, Xie and Lin (2010), whose view is that effective teaching methods and personal experiences can improve learners' creative thinking skills.

Colangelo and Davis (2011) propose role-play and brainstorming as learning activities that promote learners' creative thinking ability and imagination. They explain that a brainstorming strategy can enhance creative thinking skills because learners are required to practice thinking creatively through brainstorming. Further, the development of CTS require working conditions that enable a convenient learning atmosphere which enables the development of learners' creative thinking skills and that take into account the different learning styles and abilities of these learners (Yusnaeni et al., 2017). Training of teachers to think creatively for classroom activities is also important for developing CTS (Hosseini & Watt, 2010). According to Colangelo and Davis (2011) teachers need to practise creative motivation, awareness, attitudes and practice, and consistently involve learners in activities that require them to think creatively.

2.5 STUDIES ON TEACHING CREATIVE THINKING SKILLS

Even though the importance of teaching creative thinking skills in the classroom has long been established, studies investigating teachers' experiences of teaching creative thinking skills are scarce. Those that could be located and reviewed for this study are, Akyol and Garrison (2011), Eshet and Margaliot (2022), Larraz-Rabanos (2021), Gregory, Hardiman, Yamolinskaya, Rinne and Limb (2013), and Margaliot and Magid (2020).

In the study conducted by Akyol and Garrison (2011) it has been found that teachers should have the ability to understandCTS in order to assist in learner development on creatve thinking skills in the learning, to have the ability to develop learner perspective, to provide comfort, make the question based on developing CTS and give response to learners and to master problem-solving skills. The study further revealed that teachers should be constantly studying and developing themselves through various strategies.

The study by Eshet and Margaliot (2022) proposed that creative thinking contributes to the academic integrity of education to learners and is viewed as a creative potential skill to produce innovation and valuable solutions, therefore, teachers way of enhancing CTS in the classroom needs to be reinforced. It was also revealed that in order to improve teachers practices on developing CTS, there should be provision of educational training programs to improve creative thinking level.

Larraz-Rabanos (2021) also conducted a study and found that developing creative thinking skills in the teaching and learning process is a vital component for learners. The study focused on the development of CTS from an educational approach on how teachers apply creative strategies in the teaching and learning process. The results of this study revealed that there are relevant strategies like brainstorming, storywriting and others that were found to be appropriate for the development of CTS.

Gregory et al. (2013) study established that there is a challenge for teachers in developing CTS in the classroom as the strategies employed for classroom instruction often overlooks the importance of building creative thinking skills. However, the findings of this study show that classroom learning should offer an ideal opportunity for learners to master content knowledge and creatively apply that knowledge to indicate that CTS is crucial for success in any environment.

In the study conducted by Margaliot and Magid (2020), it was revealed the teaching of creative thinking skills requires extra effort from teachers. The study viewed that the conducive learning environment is necessary for the support of the subject matter. The findings also show that teamwork contributes in the development of CTS in the classroom as it will help teachers to understand the challenges they are likely to encounter.

Bacangallo et al.(2022) also conducted the study and found that in order to enhance creative thinking skills, teachers need to be prepared for the classroom as this will help them be innovative when developing learning activities that can be adaptive to learners needs. It was further emphasized in the study that teachers' contribution to the effectiveness of their creative ability will help them to overcome the challenges they encounter in the classroom.

Although considerable research has been conducted on creative thinking skills, the findings in this study revealed that teaching creative thinking skills is important in the classroom and should be taught for various reasons established. As it is, the information about how these skills are taught in Lesotho schools is limited. Therefore, this study investigated teachers' experiences of teaching creative thinking skills in Lesotho classrooms.

2.6 **SUMMARY**

This chapter reported on the literature reviewed. The chapter had five main sections; the introduction, theoretical framework, conceptual framework, existing studies on creative thinking skills and the summary. Under the reviewed literature, the study presented the following; the benefits of teaching creative thinking skills in the classroom, teachers' practices of teaching creative thinking skills, the factors leading to the challenges that teachers experience in teaching creative thinking, the strategies for developing creative thinking skills, and the studies on teaching creative thinking skills. The next chapter (Chapter 3) describes the methodology adopted in this study.

CHAPTER 3: RESEARCH METHODOLOGY

3.1 INTRODUCTION

This study investigated teachers' experiences of teaching creative thinking skills in Lesotho classrooms. Chapter 2, the previous chapter, reported on the literature reviewed for this study. This chapter describes the methodology used to collect and analyze data in this study. The chapter is structured in the following sections: Introduction (this section), Methodology, which includes; Research paradigm, Research design, Research methods, Selection of population for the study, Data collection methods and Data analysis methods. This study's ethical considerations and trustworthiness are also discussed and summarised.

3.2 METHODOLOGY

This section describes the methodological issues related to this study. The information is presented and discussed in the following sub-sections, research epistemology, research paradigm, research approach, research design, population and participants' selection, data collection methods and instruments, and, research ethical considerations and trustworthiness.

3.2.1 Research Epistemology: Interpretivism

This study has been framed by interpretivist epistemology. According to Alharahshel and Pius (2020) and Bhattacherjee (2012), interpretivist epistemology is the approach to social science that states comprehending individuals' beliefs, motivations, and reasoning in a social situation is critical to interpreting the meaning of the data that can be collected around the phenomenon. The interpretive further believes that the researcher is part of the research and cannot be entirely objective and removed from the research (Gray, 2014). The interpretivist believes that it is essential for researchers as social participants to understand the differences that exist among people (Saunders, Lewis & Thornhill, 2012). Similarly, different people in a society experience and understand the same reality differently and have individual reasons for their actions. Therefore, the study was guided by this interpretivism epistemology with the view that the participants were likely to exhibit different views on what is being studied.

3.2.2 Research Paradigm: Constructivism

This study adopted the constructivism paradigm. Constructivism is a philosophical paradigm that reality is not absolute because individual people construct their own ideas of realityresulting in the existence of multiple realities(Shannon-Baker, 2023). This paradigm suggests that indidividuals construct their own reality through their descriptions and narrations of their lived experiences. Further constructivism beliefs are that knowledge is co-constructed between researcher and participant (Tashakkori et al., 2021). For example, when a researcher and a participant engage in research activities they both generate data with the purpose to use to explain reality as it is experienced by the participant. Therefore, the study investigated the views of teachers about teaching creative thinking skills and how teachers understand the practices involved in the teaching of CTS. The views and the experiences of these teachers were explored through semi-structured interviews which allowed teachers to describe their own views and experiences relating to the teaching of creative thinking skills.

3.2.3 Research Approach: Qualitative

A qualitative research approach was used for this study. According to Merriam and Tisdell (2015), the qualitative approach is significantly helpful for a researcher who intends to collect a broader range of data from a diverse population. In a qualitative research approach, the researcher must understand the meanings and interpretations of the responses obtained during data collection. The participants' behaviour events or objects become essential during the analysis of the collected data (Bowling, 2014). Thus, as Babbie (2010) observed, qualitative research concerns describing and understanding rather than explaining and predicting behaviour. In addition, qualitative research approaches enable the researcher to thoroughly and appropriately analyze an issue produced by utilizing qualitative methods, and thus the participants have sufficient freedom to determine consistency (Flick, 2011).

This study was interested in understanding teachers' experiences of teaching creative thinking skills. As such, the qualitative approach was found appropriate in this study for collecting, analysing, and interpreting these experiences. Basically, the approach was advantageous in this study by providing continuous interaction between the researcher and the participants to understand their perceptions. The qualitative research in this study was beneficial for the researcher to analyze subjective meaning, events, and practices by analyzing texts and images

rather than focusing on numbers and statistics (Flick, 2014). This approach has a flexible structure, as the design can be constructed and reconstructed more (Maxwell, 2012).

Even though the qualitative approach is found appropriate for this study, it has some weaknesses. For example, it sometimes leaves out contextual sensitivities and focuses more on the meanings and experiences (Silverman, 2016); it uses a smaller sample size, and as a result, it raises the issue of generalizability to the whole population of the research (Harry & Lipsky, 2014; Thompson, 2011). Even Lam (2015) states that the small sample size results cannot be applied in other contexts. Berg and Lune (2012) also commented that qualitative research is a long, hard road with elusive data on one side.

3.2.4 Research Design: Case Study

This study used a case study design. A case study design is associated with a qualitative research approach. Cronin (2014) confirms that case study research is appropriate for qualitative research as its purpose mainly focuses on understanding complexities associated with interpersonal processes that emerge in a broader social context. Cope (2015) observes that a case study is a flexible but challenging research design commonly used in social science targeting the reader's shared understanding of everyday experience in a complex real-life situation. This is supported by Hyett, Kenny and Dickson-Swift (2014), who affirm that a case study is a research strategy that utilizes naturally existing information sources such as people and interactions between people within the scope of the case. Bless and Higson-Smith (1995) point out that case studies provide both descriptive and explanatory information and are appropriate for an in-depth investigation that focuses on one organization. Interview is data collection method often associated with case studies and were used as a source of data collection. Therefore, this study used a case study design to enable the participants to answer the research questions and provide the details of the appropriate information for an in-depth investigation of teachers' experiences of teaching creative thinking skills in the Lesotho classroom.

3.2.5 Population and Participants' Selection

The views of fifteen teachers from Hloahloeng High School were investigated in this study. According to Sugiyono (2010), the population is characterized by different features that the researcher sets to obtain specific information as such, the population targeted for this study were teachers who teaches different subjects at high school level, with different teaching experience

and who differs in age. All the particicipants selected in this study had different qualifications obtained from different institutions. Therefore, the researcher found it ideal to use this population in this study to address the pheonomenon and also as the teaching of creative thinking skills is found across all the subjects. Further, the study used a purposeful sampling technique to select the teachers who participated in this study. Johnson and Christensen (2008) describe sampling as drawing a sample from a population. As such, Arikunto (2010) defines purposive sampling as strategically selecting the participants. In this study, participants were purposefully selected and targeted to recruit teachers whose characteristics would fit the purpose of this study. The selection of the participants represented this study's population, particularly in Lesotho schools. Participants were interviewed to obtain narrative data as teachers were the relevant participants in teaching and intergrating creative thinking skills in their subjects.

3.2.6 Research Methods and Instruments

This section displayed the methods and instruments that the researcher used to collect data, and the method used in this study was semi-structured interviews. Interviews are defined as asking questions to gain knowledge and collect data from individuals (Taherdoost, 2016). In this study, the researcher preferred to use semi-structured interviews to discuss a topic of mutual interest between two or more people. Further, to obtain in-depth information about the participants' experiences by interpreting them (Doody, 2013; Bolderson, 2012). Also, these interviews helped the researcher better understand and explore the research subjects' opinions. In addition, the interview questions were open-ended so that in-depth information was collected. Therefore, this study used the semi-structured-interview method for generating data and to allow the participants to talk in any way with no restrictions.

Data Collection Methods and Instruments

According to Chiumento, Rahman, Machin, and Frith (2017), research data collection methods are activities or tools designed to gather information in research undertaken. Cresswell (2014) further attests that data collection research methods are the strategies or techniques used to collect information in the study. Thus, the methods used in this study were semi-structured interviews which the researcher found most suitable for investigating the phenomenon. At the same time, the study aimed at identifying the activities or tasks that teachers give learners to

develop creative thinking skills, and examine the challenges that teachers experience in developing creative thinking skills to the learners. As well as identify ways to equip teachers on how to deal with such challenges through the semi-structured interviews.

Semi-structured interviews. The data instrument used in this study to collect data was semi-structured interviews. This type of interview is flexible and allows the researcher to probe and seek knowledge from the participants responses. A semi-structured interview is a data collection technique that asks participants open-ended questions to explore their responses and the topic of interest (David & Sutton, 2004; Doody, 2013). In addition, since semi-structured interviews are considered open-ended, they are flexible and allow supplementary questions to be asked to guide the discussions and allow participants to speak liberally about what is in their hearts (Taylor, 2015). Therefore, in this study, the semi-structured interviews provided the interviewer and interviewee with opportunities to discuss and explore the topic areas in more detail (Magald and Berler, 2020). The method also provided the researcher an opportunity to explore responses, which were significant and beneficial for the focus of the study.

Data collection

Before the study was conducted, the Faculty of Education in the National University of Lesothogranted the researcher permission to conduct the research. Aletter of introduction was given to the principal of the selected school. Since data was collected in the school where the researcher works, the principal just informed the teachers about the purpose of the research allowed the researcher to select the potential participants for the study. All the participants were recruited after school and the researcher made arrangements for the interviews.

Before the interview was carried out, the researcher explained the purpose of the study to each participant and informed them about their freedom to withdraw at any time during the interview if they wished to. Then the participants were asked to sign a letter of consent as an indication that they were willing to participate in the study. During the day of interview, the participants were given the forms to fill in the provided spaces for answers. All the participants were given an hour to answer all the questions which were asked in English.

Data analysis methods and instruments

Data analysis is preparing and organizing data collected for a study (Cresswell, 2013 & Rakotsoane, 2012). According to Cresswell (2012) and Rakotsoane (2012), the process can start with moving from particular pieces of information to more detailed information. For data analysis in this study, the researcher used a thematic analysis approach as an appropriate and powerful method to seek and understand a set of experiences, thoughts, or behaviours across data (Braun & Clarke, 2012). According to Clarke and Braun (2017), thematic analysis consists of six steps, and as such, in this study, the researcher used this approach to analyze data. The researcher used this approach to analyze data. The following were the steps that the researcher followed when analyzing data:

Step 1: Becoming Familiar with the Data

The researcher familiarized herself with the data set, which encompasses interviews. At this stage, it was helpful for the researcher to make notes and jot down early impressions.

Step 2: Generating Initial Codes

In this study, the researcher collected raw data and transcribed it to generate codes. In this regard, the data was organized meaningfully and systematically to generate themes by combining familiar responses. The researcher first made codes from the participants' responses by grouping the common responses and naming them. The researcher further classified the common responses with similar issues together.

Step 3: Searching for Themes

As this step involves examining the coded data, the researcher extracted the information from the responses to look for potential themes of broader significance. The researcher constructed themes through analysis, and the themes that were generated were as follows:

- 1. Teachers' interpretation of creative thinking skills (CTS)
- 2. Teachers' practices of teaching CTS in the classroom
- 3. Factors leading to the challenges that teachers experience in teaching CTS.
- 4. How teachers are supported to teach CTS
- 5. Teachers' general views about teaching CTS in Lesotho classrooms

Step 4: Review Themes

In this phase, the researcher reviewed, amended, and developed the initial themes identified in step 3.

Step 5: Defining and Naming Themes

This is the final modification of the themes aimed at identifying the core of each theme. In this step, the researcher engaged in deep analytic work involved in thematic analysis, the crucial shaping of analysis into its fine smooth detail. Then selected extracts to present or analyze themes then set out each theme's story.

Step 6: Producing the Report

The researcher produced a report on the findings in the final analysis phase. That is, the findings from the analysis of the results were transcribed in line with the major research questions.

3.2.7 Ethical Considerations and Trustworthiness

This section addresses the ethical considerations and trustworthiness of the participants. However, the study was conducted at Hloahleng High School (pseudonym), and through communication with the researcher and the school's principal; the permission to collect data was granted.

Ethical Considerations. To ensure that a study adheres to basic ethical requirements and that no harm was done during the research process, and there was adherence to certain principles. Furthermore, these include doing no harm to the participants and ensuring that the rights of the participants are respected during research (Creswell, 2012; Louw et al., 2014 & Rakotsoane, 2012). To ensure that this study was ethical, the researcher initially asked for permission from the National University of Lesotho and from the principal of the selected school. Also, before the participants for this study were recruited, the teachers were informed about the purpose of the study and how their information is going to be used in the study. The participants were also informed that they may withdraw if they wished.

Doing no harm. All the participants were ensured that nothing would cause physical or emotional harm. Their preconceptions or opinions would not be interfered with during the collection process. The participants were also ensured that the information provided was confidential and that the findings would be anonymous.

Respecting the rights of the participants. Theresearch protocol was followed to ensure that all the participants were informed about the study and their participation. They were also informed that their dignity, privacy, and interests would be respected. With this information, the participants were asked to fill in a consent form indicating their participation decision was voluntary, ensuring that their rights and time were respected.

Trustworthiness.Trustworthiness of the study refers to the level of trust in data, interpretation, and methods used to ensure data quality in a study (Polit & Beck, 2014). According to Kumar (2011) and Connelly (2016), research trustworthiness postulates that the researcher's words can be trusted as representing truthful statements. On the same note, Yin (2011) indicates that truthfulness is vital in qualitative research. The aspects of trustworthiness include credibility, transferability, dependability and confirmability. As a result, the study had followed the following strategies:

Credibility: The credibility of a study is is about whether the study's findings are accurate and could be trusted and believed in (Polit & Beck, 2014). In this study, the study credibility was ensured by using a technique called member check for members to check and certify information before analysis. As such, the collected data was sent back to the participants to verify and confirm that the transcribed data reflected their experiences. This was also ensured through a prolonged engagement with data to identify the overlapping themes and to eliminate the contradictions (Stahl & King, 2020). All the participants were satisfied that the interviews represented enough content that was presented in the interviews.

Transferability: Transferability is how the researcher can demonstrate that the study's findings apply to the contexts (Kumar, 2011; Thomas & Magilvy, 2011). To ensure transferability in this study, the researcher provided sufficient information about how data was collected and analyzed. Also, the researcher provided a detailed description that clarifies all the processes that were taken during data collection. The interview method has been tried and used before and have been found reliable to produce information intended if applied in contexts similar to the one in which this study was conducted.

Dependability: Dependability refers to the extent to which a study can be repeated by other researchers and yield the same results (Kumar, 2011 & Merriam, 2009). In this regard, Ary (2010)

advocates that the researchers can divide data, independently analyze it, and then compare the results. To ensure dependability in this study, the researcher connected the results and data provided to observe if they gave the same results.

Confirmability: Confirmability concerns whether the research findings could be repeated with the same participants to confirm neutrality and avoid bias (Anney, 2014). The study paid attention to coherence in terms of the information gathered and the interpretation of data (Trochim & Donnelly, 2008). In this study, the researcher documented the procedures by conducting a data review for checking and re-checking data throughout the study. Additionally, the researcher achieved confirmability by ensuring transparency by using colleagues as critical readers.

3.3 **SUMMARY**

This chapter presented the methodology that was used in this study. The chapter begins with and introduction, research epistemology, constructivism research paradigm, qualitative research approach and case study as research design. The study further explained methods of data collection and instruments, data analysis methods and instruments, ethical considerations and trustworthiness. The next chapter (Chapter 4) presents the findings of this study through the data collection methods used in this study.

CHAPTER 4: FINDINGS

4.1 INTRODUCTION

This study aimed to investigate teachers' experiences of teaching creative thinking skills in Lesotho classrooms. Data that was collected for this study was about (1) how teachers in Lesotho classrooms interpret creative thinking skills and their views on the benefits of developing learners' creative thinking skills; (2) teachers' practices of teaching creative thinking skills; (3) factors leading to the challenges that teachers experience in teaching CTS; (4) how teachers are supported to teach CTS; and (5) teachers' general views about teaching these skills in Lesotho classrooms. In the previous chapter (Chapter 3), the methods that were used to collect data were described. This study's data was gathered through semi-structured interviews and analyzed using a thematic approach. This chapter presents the findings of this study. The chapter is structured as follows: Introduction (this section), The participants' profile, Presentation of results, and Summary.

4.2 THE PARTICIPANTS' PROFILE

The participants in this study were selected from one high school in Lesotho. How these participants were selected is detailed in Chapter 3. For confidentiality purposes, the identities of the school and the teachers have been pseudonym, the school as Hloahloeng High School (pseudonym) and teachers as Teacher (1) to Teacher (15). The profiles of these teachers are provided in Table 4.1. As indicated in this table, the participants in this study were males (08) and females (07). All the participants were from one school Hloahloeng High School (pseudonym). The participants' ages ranged from 40 to 65 years, with seven females between 40 to 59 years old and eight males between 42 and 65 years old. Their years of teaching experience ranged from eight to 38 years. As reflected in the table, the participants had different levels of highest teaching qualifications, which were Secondary Teacher Certificate (STC) from Lesotho College of Education (LCE), which was formerly known as National Teachers' Training College (NTTC); a Diploma in Education Secondary (DES) from LCE; a Diploma in Computer Studies from Bethel College; Bachelor's degree in Education (B.ED) from the National University of Lesotho (NUL). Only one

participant had less than 10 years of teaching experience. The rest of the participants had 15 years and more than 15 years of teaching experience. All the participants were teaching from Grade 8 to Grade 11.

Table 4.1: Summary of participants' professional profile

Profile	Ag	Gen.	Highest	Institution	Special.	Exp.	Teaching
	е		Qual.				Grades
Teacher 1	44	F	STC/BED	NTTC/NUL	Dev. Studies & Eng. Lang.	20	8-11
Teacher 2	43	М	DES/BED	LCE/NUL	Commercials	15	8-11
Teacher 3	42	М	BED	NUL	Eng. Lang.& Dev.Studies	16	8-11
Teacher 4	46	F	DES/BED	LCE/NUL	Maths.& Chemistry	16	9-11
Teacher 5	59	F	DIP. EDU	NUL	Agriculture & Science	33	9-11
Teacher 6	43	М	BED	NUL	English Lang.& Dev. Studies	15	9-10
Teacher 7	40	F	BED	NUL	Accounting, Econ.& Bus.	08	9-10
					Studies		
Teacher 8	51	М	BSC/PGDE	NUL	Biology, Chemistry & Maths	29	8-11
Teacher 9	65	М	DIP/ BSC	NUL	Science & Agriculture	38	9-10
Teacher 10	58	М	DIP/ICDL	BETHEL	Computer Studies	20	8-11
Teacher 11	47	F	BED HON.	NUL	Eng. Literature & Dev.	25	8-11
					Studies		
Teacher 12	40	М	DIP in	NUL	Agric & Junior Science	13	8-11
			AGRIC ED				
Teacher 13	47	М	BED	NUL	Physics and Maths	22	8-11
Teacher 14	42	F	BED	NUL	Eng. & Dev. Studies	16	9-11
Teacher 15	36	F	BED	NUL	Accounting & Bus. Studies	10	9-11

4.3 PRESENTATION OF THE RESULTS

This section presents the data collected from the interviews of eight teachers described in the previous section. These teachers were asked about their (1) interpretation of creative thinking skills (CTS) and their views on the benefits of developing CTS; (2) teacheres' practices of teaching CTS, (3) factors leading to the challenges that teachers experience in teaching CTS; (4) how teachers are supported to teach CTS, and (5) views about teaching CTS in Lesotho classrooms. A

thematic approach was used for the analysis of this information. Five themes emerged from this analysis. The findings from the analysis are presented in line with these themes.

4.3.1 Teachers' Interpretation of Creative Thinking Skills

Teachers were asked to define creative thinking skills and to outline the benefits of developing learners' creative thinking skills. This information is presented in two sub-sections: the definition and the benefits.

Definition of creative thinking skills. The participants in this study understood creative thinking skills as the ability to bring new ideas to generate and provide innovative solutions to the current problem. For example, Teacher 2 described creative thinking skills as:

The ability to think critically and analyze the information presented to come up with a solution to the problem. In simple terms, I consider it to be the ability to think outside the box.

Only one participant interpreted CTS differently. Her interpretation linked creative thinking skills with creative art and described it as an activity that engages learners in visual arts. Teacher 7 explained this by saying, "Creative thinking skill is the work of art whereby learners create a new product out of their imaginations." Apart from this participant, all seven participants in this study had the same interpretation of CTS.

The participants' way of thinking about creative thinking skills appeared to be in-line with other scholars such as Daud et al. (2012), Gregory et al. (2013), Scott (2015), and Khoiri (2019), who view creative thinking skills as an ability to make or bring to existence something new or bring a solution to a new problem. These scholars also observe that creative thinking skills are part of cognitive learning that aims to solve problems. They believe creative thinking is needed in the teaching and learning processes to help learners learn skills that will help them deal with and find solutions for the complex problems they encounter in their different learning subjects and everyday life situations.

Benefits of developing learners' CTS. When the participants were asked about their views on the benefits of developing learners' creative thinking skills in teaching and learning, they indicated that it is vital for learners to be taught CTS in the classroom for various reasons that include these skills:

Contribute to learners' humanity. The participants established that it is essential for learners to be equipped with this 21st century skill as it contributes to their humanity by promoting individual well-being in personal and professional achievements.

Increase emotional intelligence. The teachers who participated in this study further highlighted that learning CTS increases emotional intelligence, which helps learners to deal with stressful situations. The participants' views in this study align with that of Adams (2019), who established that teaching creative thinking skills responds to the needs of the learner's personality to find new ways of addressing issues like stress. This view was reflected in Teacher 6's response:

Creative thinking skills should be promoted in our classrooms as valuable because they are essential for promoting individual well-being in personal and skilled achievements. It also enables learners to resolve challenges that may emerge, like stress.

Provide a learner with resilience at the workplace and in the community. The participants also expressed that learners are affected every day by the demands of rapid change in the workplace and the community. Accordingly, these participants confirmed that thinking creatively could improve working practices and help develop and implement innovative solutions. They further expanded that learning this skill in a community brings about different perspectives that lead to acceptance of individual differences. This view has been confirmed by Nuzliah (2015), who stated that it is often difficult to solve problems and create new ideas to overcome community and workplace challenges without creative thinking ability. This is how Teacher 8 explained this view:

After dropping out of school, most of my learners encounter challenges in their communities. They cannot survive their everyday life obstacles like accepting individuals' characters. For those who have been lucky to finish school, when they are at their workplaces, they are not productive in providing businesses with new ideas, which becomes the worst challenge.

Some participants further expanded that CTS provide learners with the opportunity to develop skills that make them productive in their workplace and their communities. For example, this is how Teacher 5 expressed this view:

The opportunities provided to learners when they are taught creative thinking prepare them to be more productive in their future workplace and be better positioned for success. In addition, in the community, they can interact with other community members to resolve conflicts.

Provide learners with opportunities to gain knowledge and experienceneeded to solve their Problems. This view is reflected in Teacher 6's response:

Creative thinking skills allow learners to extend their imaginations and think ahead of their present to solve a problem. It further helps learners to bring different perspectives and new ideas into the classroom.

Thus, this view about the learners' opportunity to gain knowledge from being taught CTS in the classroom appeared to be in line with the views of Hobbs (2010), who affirms that creative thinking skills help learners to learn independently and with minimal assistance to critically think about their problems and gain knowledge and experience to solve problems.

Promote Learners' Problem-Solving and Interpersonal Skills. This was particularly mentioned by Teacher 2 when she said, "It promotes problem-solving, develops interpersonal skills and motivates learners." Teacher 1, who said, "Teaching creative thinking skills in the classroom helps learners to solve problems and create interpersonal relations among themselves", supported Teacher 2

Assist Learners to Develop Innovative Ideas. Teacher 5 elaborated on how important it is for learners to develop innovative ideas that bring new perspectives and how CTS effectively assist learners with this. This is how she responded:

Learners need to develop creative thinking skills because it will help them to perceive patterns that are not obvious and bring new perspectives into the classroom. They will also become innovative in their approach to solving problems.

Enable learners to expand their knowledge. This is how Teacher 3 expressed this view:

Teaching creative thinking skills in the classroom enables learners to discover unknown concepts from the known. Learners with creative thinking skills can apply any concept to the relevant situation.

Generally, these results show that the teachers who participated in this study discovered that it is essential for learners in Lesotho classrooms to be taught CTS for various reasons. This participant's view coincides with the views of Zarkasyi (2015), who indicated that the learning of creative thinking allows learners to gain knowledge and experiences and to discover, recognize, and solve problems with multiple techniques. The participants further pointed out that teaching CTS can apply to all different school subjects. (in what ways, an elaboration would be interesting to know. Please include more information).

4.3.2 Teachers' Practices of Teaching Creative Thinking Skills

The participants in this study claimed that they employ or utilize different teaching methods to develop CTS in the classroom. Teachers indicated that they use discovery, brainstorming, experiential learning, group discussion, games, and puzzles for teaching CTS.

Discovery method. The participants suggested that a discovery method was one of the methods they used to promote learners' CTS in the classroom. They claimed that the reason they preferred this method is that it can effectively develop the learners' CTS. For example, Teacher 4 justified this method by saying:

I design lessons that require learners to discover things for themselves by giving them concepts that require them to think creatively. I also employ practical strategies to enhance the effectiveness of learners' creative thinking skills.

Brainstorming method. The participants in this study highlighted that they engage learners in brainstorming activities as a way of developing their CTS. They indicated that this activity allows the learners to exercise their imaginations. For example, Teacher 3 response was that she uses brainstorming because it is a strategy that can enhance creative thinking ability. This teacher explained that this method requires learners to think creatively. For example, this is how this teacher explained:

In my English Language class, I provide learners with stories and ask them to create their own stories similar to the story narrated. In the process, I allow them to brainstorm ideas and debate them.

Teacher 6, who described how she engages learners in the brainstorming activity, supported this view. This is how this participant explained:

I always give learners a couple of tasks to conduct research and organize brainstorming sessions relating to their research. Sometimes I even ask them to present their findings to others.

How teachers use the brainstorming method to promote CTS appeared to be in line with Anco and Unluer (2010), whose view is that brainstorming is one of the learning activities that can promote the creative thinking ability and imagination of learners.

Experiential learning. The participants designed experiments in their subjects as a means of developing CTS. This helps learners bring as many solutions as possible to a given question and promote their thinking ability. These findings are aligned with the views of Lucas and Spencer (2017), who observed that experiential learning promotes learners' creative thinking skills because they are practical and motivate learners to think creatively. They further expanded on experiential learning as a method that helps learners to make informed decisions. Teacher 8 explained this by saying:

I design experiments that allow learners to carry out steps based on the guided concept. After they are given a chance to design their experiments, they are given a set of conditions to follow in coming up with possible answers to a given question, and these help my learners to move from a problem to a solution and to make informed decisions.

Games and puzzles learning. The participants indicated that they engage their learners in instructional games and puzzles to promote CTS in the classroom. They further explained that through games and puzzles learning, learners cultivate their thinking ability, so their learning of CTS grows naturally. This view is reflected in Teacher 8's response who said:

The methods that I normally use involve playing puzzles and games to enhance creative thinking skills in my class. As I teach Mathematics, I engage learners in crossword puzzles to teach concepts like addition, subtraction and multiplication. In this activity, learners apply their creative ability to solve problems. In addition, learning Mathematics using puzzles encourages learners' interest in the subject.

Gao (2015), who established that games and puzzles encourage learners to think creatively and help them absorb details from concrete imaginations, confirmed this view.

Some participants further established that games and puzzles provide a mental challenge that helps learners to develop their CTS. These games and puzzles allow for trial and error so learners can learn from their mistakes and develop new ideas to solve complex problems. This is how Teacher 2 clarified this point:

I give learners tasks that involve them applying their creative ability. I request them to make a picture using a puzzle, and then they will put together different pieces together to complete it. This type of game promotes creative thinking since learners need to pay attention to small details to complete the puzzle.

Discussion method. The participants used the discussion method to develop CTS in their classrooms because it pushes learners to build creative thinking as they continually keep incorporating one another's ideas into the solution. For example, this is how Teacher 4 explained:

I usually use the discussion method in the classroom to develop creative thinking. I give learners a task that forces them to listen to one another's ideas and suggest their ideas to build on them. This helps them to get experience in developing creative thinking.

This view of the participants aligns with Huang, Wang, Chen, Yu and Chou (2017), who observed that learners work cooperatively during discussions and acquire experience, and creative skills to cope with complex problems. The participants in this study further claimed that they engage learners in the group discussions as a way of promoting their CTS. For example, this is how Teacher 1 expressed this view:

I usually assign learners to research a concept and present it in groups so that everyone in the group is given a chance to engage in the discussion to come up with ideas and present them.

4.3.3 Factors Leading to the Challenges that Teachers Experience in Teaching CTS

The findings in this study indicated that many teachers encounter challenges when teaching CTS in their classrooms. Many of them claimed that their way of teaching CTS in the classroom is not as successfully as they wanted. Others pointed out that they do not teach these skills at all. They mentioned factors such as: lack of skills and knowledge, lack of resources, the calibre

of learners, and limited time being responsible for the challenges they encounter in the classroom.

Teachers' lack of knowledge and skills. Six of the participants explained that they are not well informed on how to teach creative thinking skills in their subjects. As a result, they hardly give their learners activities that require them to think creatively. They explained that they do not possess adequate skills needed to implement this skill. This is how Teacher 5 explained:

I do not give learners activities that promote creative skills. Also, Ido not prepare my lessons to incorporate creative thinking skills in my subject. This is because I am not familiar with creative thinking skills. In short, I do not know anything about these skills.

Teacher 7 had the same view as Teacher 5. He indicated this by saying, "I am not a critical thinker myself. Therefore, I have a problem to instill the skill to learners. I also lack knowledge of the skill."

Teacher 1's view was also the same as the views of Teacher 5, and Teacher 7, that teachers lacked sufficient skills to teach CTS. This teacher said:

As a teacher, I lack knowledge and adequate skills; at times, I just research and give learners the findings without demystifying the concepts to them. I need to acquire sufficient skills to teach creative thinking first before I can attempt to teach them in my class.

Some participants indicated that because of the limited knowledge they have about CTS, They have low self-esteem about teaching these skills. They explained that they find it challenging to engage learners in creative thinking activities because of their low confidence. Suciawati (2019) has reported the same finding that the main challenge that many teachers face in teaching CTS is poor self-confidence, low self-esteem, and a feeling of helplessness, which are caused by these teachers' limited knowledge. This is how Teacher 7 expressed this view:

I am not a critical thinker myself, so I have a problem to instill this skill to learners. In addition, I have low self-esteem, which might affect my learners' creative behaviors and their potential to produce creative outcomes.

As established in this study, some teachers cannot even identify strategies that can effectively develop learners' CTS due to lack of knowledge. This was particularly pointed out by Teacher 6

who said, "I do not employ creative teaching strategies in my subject because I am not very conversant with them." In support of this claim, Teacher 2 said:

As a teacher, I do not utilize strategies that promote creative thinking because I do not know them. I normally teach content without giving learners chance to think for themselves, for example, most of the time my teaching is teacher- centered.

Lack of resources. The participants further indicated that lack of resources is another factor that influences how they develop CTS in the classrooms and this factor makes it difficult for themes to teach the skill. For example, this is how Teacher 4 expressed this view:

Schools' lack of resources that are required to make creative teaching skills easy to be taught and developed is a problem. Some learners are too slow to think and they need resources that can assist in developing of creative thinking skills.

Teacher 5 had the same view as Teacher 4 by indicating that the teaching of CTS is not practiced because schools lack creative teaching materials. This was reflected in the Teacher 4 response, which was, 'I am not able to teach learners about creative thinking skills because there are no materials that would assist me to develop and promote this skill in my classroom.'

In this case, established that lack of resources in schools influence the way they teach CTS in the classroom and that makes it difficult for them to develop this skill in the classroom. This view of the participants is in-line with those of Eragamreddy (2013) and Saadi (2016), who observed that schools' lack of facilities for learners to develop the ability to think creatively and independently hinders the promotion of creative thinking in the classroom.

Calibre of learners. Teachers stated that learners' lack of exposure is another factor that influences how they teach CTS. This factor makes it difficult for teachers to promote and develop this skill in their classrooms because, in other classes, learners are not exposed to situations that could make them to use their imaginations and think creatively. For example, this is how Teacher 2 clarified this point:

Most learners in my classroom are too slow to think out of the box. Whenever they are given task or activity to think about or stretch their minds, they provide the same

responses of the things they are familiar or of things happening around them and in that case I become a dictator and think for them

Teacher 3 supported this point by saying, "Learners lack exposure to different things especially concept based ideas." In addition, Teacher 5 also believed that:

Learners are lazy to think and they are not interested to learn new things due to lack of exposure. What I mean is they are not exposed to situations in which they may be obliged to practice to use their imaginations and think creatively.

Generally, the participants pointed out that their learners cannot think creatively because they lack exposure to different situations and it becomes difficult for them to extend their imaginations. They are too slow to think, a result they cannot think creatively. The view of these teachers is aligned with the views of Chang (2013) who established that teachers find it difficult to develop CTS in the classroom as learners have limited knowledge and exposure.

Time factor. The participants expressed that limited time is another factor that influences the way they teach CTS. They highlighted that the teaching of CTS is time-consuming as a result, they are not teaching this skill in their classrooms. For example, this view is reflected in for Teacher 3 who said:

I am overloaded with work and I do not have time to prepare for teaching creative thinking skills since I have to prepare for other subjects that I teach. I am also overloaded with the classes that I teach, therefore the time to prepare for creative thinking would consume a lot of time, which I don't have hence I don't teach creative thinking at all.

Some participants complained about the strategies that are used for promoting CTS in the classroom. Their view is that these strategies require a lot of time and effort for planning and preparation. They explained that this makes it impossible for them to teach CTS in their subjects as it requires a thoughtful planning and preparation. This point has been expressed by Teacher 6 who said:

I am overloaded with the classes that I teach therefore, the time to prepare for creative thinking would consume a lot of my time. Further, my workload will not allow me to have enough time to design activities that would engage learners in developing creative thinking.

This has been affirmed by Beghetto and Kaufman (2012) that teaching creative thinking skills requires intensive planning and design of learning tasks that foster learners' creative thinking. Hence, this puts a lot of pressure on teachers, as it requires a lot of time for preparation. Tangaard (2011) also established that time is one of the factors that influence teachers' reluctance to engage learners in creative thinking.

4.3.5 Teachers Views on how they are supported to Teach CTS

The participants in this study expressed their views on the existing situation in their schools in relation to how they are supported to teach CTS in their classrooms. They also showed their concerns that the current conditions in their schools do not allow them to teach this skill.

In view of how they are supported, the participants indicated that, currently they are not teaching CTS in their classrooms either by their schools or the Ministry of Education and Training. From their point of view, neither the schools or the Ministry of Education and Training provide teaching resources and training workshops that can assist them in anyway. The view of these participants is that both the Ministry and the schools are silent about the teaching of CTS and, as a result, CTS is not taken into consideration across all the subjects. These teachers explained how this frustrates and demotivate them to teach these important skills. For example, Teacher 4 explained this by saying:

In my school, teaching of creative thinking skills frustrates us because the school does not provide us with resources, let alone holding mini seminars to equip us with the skills that develop creative thinking. Further, the Ministry of Education has not offered any training on the teaching of this strategy. This says that to us no one remains accountable for implementing this skill in schools.

Teacher 6 who had the same view as Teacher 4 also explained:

As teachers, currently we are not teaching creative thinking skills because the management in our school has not made any effort to assist us to teach this skill. In this case, creative thinking is not taught at all. In addition, the Ministry also has never met

schools halfway by providing resources or a resource person to help in the teaching of creative thinking skills.

The participants also complained that even the management in their school is not taking any responsibility to guide them on how they can teach and develop CTS in their classrooms. In support of this complain, Teacher 3 said:

I do not feel happy to teach creative thinking in my classroom because no one is giving me guidance on how to develop this skill during my teaching. Even my Head of Department who is responsible for classroom instruction is not helping me when the need arises, hence I do not teach this skill.

Some views that have been expressed by the participants in this align with Adams's (2019) finding that teachers are not sufficiently supported by professional education organizations.

4.3.6 Teachers' General Views about Teaching CTS in Lesotho classrooms

When teachers were asked to give their general views about teaching CTS in Lesotho classrooms, they believed that CTS are important skills and they need to be taught to every learner. They further suggested that this skill could be implementable in the classrooms when teachers have necessary skills and when the resources are available. They also indicated even in Lesotho classrooms, the teaching of CTS can be effectively implemented if the Ministry of Education and the schools could assist with the following; training workshops, teaching resources and resource teachers and facilitators, and conducive learning environment.

Training workshops. The participants in this study indicated that they teach (or do not teach) CTS in their classrooms in a limited way because they lack knowledge and adequate skills necessary to teach the skill. They suggested that if they could be exposed to the training workshops organized by the Ministry of Education or other stakeholders in the system they will be able to develop the necessary skills and will be motivated to teach CTS. Hosseini and Watt (2010) support the view that teachers need to be trained with the intention to equip them with skills that would enable them to think creatively and to promote CTS among their learners. For example, this is how Teacher 2 showed this view:

I teach creative thinking skills in the classroom but in a limited way because I lack knowledge and the necessary skills that can help me to develop this skill in the classroom.

Therefore, I need training workshops that will equip me with the techniques to think creatively and teach the skill

The participants further explained that the training workshops would effectively expose them to different teaching methods that are associated with the teaching of CTS. This is how Teacher 8 expounded on this point:

The schools and the Ministry of Education should provide us with regular training workshops whereby we can be taught and be exposed to different teaching methods that would help us to develop creative thinking in learners.

Teaching resources. According to the participants, for CTS to be successfully taught in Lesotho classrooms, there should be appropriate teaching resources that will make this strategy effective. From their viewpoint, as indicated earlier in this chapter, schools' lack of resources hinders the promotion of CTS in the classroom. This was particularly emphasised by Teacher 7 who said:

If the appropriate teaching resources were there, teachers would not overlook the teaching of creative thinking skills in classroom because the availability of resources would limit no excuse for teachers in not teaching and developing the skill.

Supporting this argument, Teacher 1 pointed out that:

It is difficult to teach creative thinking skills in my subject. As an English teacher, I sometimes teach stories that I want my learners to visualize on screen, I need my school to assist me with computers and Wi-Fi so that learners may have access to the internet to watch video clips that are relevant to the development of creative thinking skill.

Resource teachers or facilitators. The participants in this study believed that the Ministry of Education or the schools should employ resource teachers or facilitators who are knowledgeable in teaching this strategy. Their view is that through the guidance of this human resource, teachers could acquire relevant knowledge and skills. According to these participants, the Ministry of Education's engagement in providing schools with resource teachers who have expertise in teaching creative thinking skills should be mandatory.

Conducive learning environment. The findings in this study indicated thatthe participants believed that schools' creation of a conducive learning environment could also make it easy for teachers to teach CTS in Lesotho classrooms. Their argeument was that a conducive learning environment supports intrinsic motivation for both teachers and learners. According to these teachers, such environment can help teachers and learners to explore the idea of creative competence. This is how Teacher 1 explained:

Our schools should create a conducive learning environment for both teachers and learners. This can help them to be creative and develop a vision of how ideas could be applied in different contexts, resulting in intrinsic motivation.

Few participants in this study beelived that the government should be fully involved to see that the teaching of CTS is possible in Lesotho classrooms. Their argument was that despite its benefits, the teaching of CTS is time-consuming and requires a lot of time for preparation even where resources could be easily available. Teacher 5 explained this by saying:

I do not encourage schools to implement creative thinking skills in classrooms because it consumes a lot of our time for preparation and implementation.

They advanced their argument that, if there is an insistence that this skill should be taught, then the Ministry of Education should start by piloting it first in some schools in order to gauge the materials and the equipment needed for teaching it. For example, this is how Teacher 6 clarified this point:

In order for us to teach creative thinking skills in our school, the ministry should select a few schools that would be used as pioneers in the development of creative thinking in the classroom. This will make it easy for the ministry to design appropriate teaching resources for teaching this skill.

The general observation about these findings is that, the participants believed that CTS could be practiced in Lesotho classrooms if they could receive support from the Ministry of Education and their schools. From their point of view, training teachers and employing resource teachers could help in equipping teachers with necessary skills to teach and develop CTS in the classroom. They further observed that the schools should create a conducive learning environment for both

teachers and learners to develop this skill. These views appeared to be in-line with the views of Yusnaeni et al. (2017) who observed that learners' unique creative thinking abilities require conditions that pay attention and involve each individual learner in the learning experiences that suit their abilities and learning styles, and the well-informed teachers who are supported in all areas of teaching creative thinking.

4.4 SUMMARY

This chapter presented the findings of teachers' experiences of teaching creative thinking skills in Lesotho classrooms. This information was presented into five main sections: interpretation of creative thinking skills students' creative thinking skills; teachers 'practices of teaching creative thinking skills; factors that that influence the challenges that teachers experience in teaching creative thinking skills; how teachers are supported to teach creative thinking skills; and teachers' views about teaching these skills in Lesotho classrooms. About their interpretations, the results revealed that teachers have common understanding of creative thinking skills except for one who linked it to creative art. These further expressed their views on the benefits of developing CTS. The results show that some teachers claimed to teach this skill but in a limited manner. The factors that lead to the challenges that make it difficult for teachers to teach CTS were revealed in this chapter, such as (provide one example from the chapter). The findings also indicate how teachers are supported to teach this skill. The views of these teachers about teaching this skill in Lesotho classroom have also been revealed in this chapter. In the next chapter (Chapter 5), the discussions, conclusions and recommendations made by this study are presented.

CHAPTER 5: DISCUSSIONS, CONCLUSIONS AND RECOMMENDATIONS

5.1 INTRODUCTION

This study investigated teachers' experiences of teaching creative thinking skills (CTS) in Lesotho classrooms. This qualitative study collected information using semi-structured interviews and analyzed using a thematic analysis approach that included:

- 1. Teachers' interpretations of creative thinking skills
- 2. Teachers' practices of teaching creative thinking skills
- 3. Factors leading to the challenges that teachers experience in teaching CTS
- 4. How teachers are supported to teach CTS in the classroom
- 5. Teachers' general views about teaching creative thinking skills in Lesotho classrooms

The previous chapter (Chapter 4) presented the findings of this study. This chapter (Chapter 5) presents the major findings of this study and discusses the results, draws conclusions, highlights the study's limitations, and finally outlines the recommendations made by this study. Thus, this information is presented as follows:

5.2 MAJOR FINDINGS

The discussion of the results in this study are presented in five major themes. These discussions are structured as: 1) teachers' interpretations of and views on the benefits of developing CTS in the classrooms; 2) teachers' practices of teaching CTS in classrooms; 3) factors leading to the challenges that teachers experience in teaching CTS; 4) how teachers are supported to teach CTS in the classroom; And 5) teachers' general views about teaching these skills in Lesotho classrooms.

5.2.1 Teachers' Interpretation of CTS

Teachers in this study showed their understanding of CTS. They interpreted it as the ability to bring new ideas to generate solutions to existing problems. They further expanded that it is the ability to produce something new through imagination by giving a solution to a problem. Only one teacher in this study interpreted CTS differently. Her interpretation linked creative thinking

skills with creative art, describing it as an activity that engages learners in visual arts. Thus, generally, teachers' way of thinking about creative thinking skills appeared to be in line with the views of scholars like Tendrita et al. (2016), Cintia, Kristin and Anugraheni (2018), and Maksum, Widiana, and Marini (2021) who view creative thinking skills as a thinking ability that can produce the variation of possible responses to solve problems. These scholars also believe that creative thinking is needed in the teaching and learning process to help learners develop skills that assist them in dealing with and finding solutions to complex problems.

5.2.2 Teachers' Views on the Benefits of CTS in the Classroom

In this study, teachers discovered that teaching CTS in the classroom is essential for learners for many reasons. They indicated that it is vital for learners to be equipped with this 21st century skill as it contributes to their humanity by promoting individual well-being in personal and professional achievements. These teachers further expressed that learning CTS increases emotional intelligence, which helps learners deal with stressful situations. Adams (2019) and Plucker, Runcoab and Siminsen (2020) believe that teaching creative thinking skills responds to the needs of learners' personalities to find new ways of addressing complex issues and facilitates learners' personal academic growth.

Additionally, teachers are of the view that learners, because learners are affected every day by the demands of rapid changes in the workplace and the community, they need this skill. They highlighted that learners need this skill because it improves their working practices and helps them develop and implement innovative solutions. These teachers also believe that because of this skill, learners have different perspectives that lead to accepting individual differences. This view is supported by Nuzliah (2015), whose belief is that, without creative thinking ability, it is often difficult to solve problems and create new ideas to overcome community and workplace challenges.

Developing CTS in the classroom helps learners to produce innovative ideas that assist them in overcoming complex problems and bringing about innovation. According to this study, CTS enable learners to translate unknown concepts into known concepts to solve problems. This process allows them to extend their imaginations and think ahead of their present situation to solve a problem.

Because of their views of these benefits, teachers in this study believe that CTS should be taught to help learners develop abilities that will allow them to deal with challenges in everyday life situations and find solutions to them. The teachers' beliefs appear to be supported by Carson (2010), who believes that teaching this skill promotes learners' ability to cope, resolve complex problems, adjust and succeed in their personal lives, education, business, and society.

5.2.3 Teachers' Practices of Teaching CTS in the Classroom

This study claims that teachers use different teaching methods that develop CTS in the classroom. Some of the methods they listed include discovery, brainstorming, experiential learning, discussion, and games and puzzles. These teachers assert that where these methods are appropriately planned, they effectively develop learners' CTS. This is because they allow learners to exercise their imaginations. Anco and Unluer (2010) have also observed that teaching methods that engage learners in activities requiring them to solve problems promote their creative thinking ability and imagination. For example, they argue that experiential learning helps learners bring about as many solutions as possible to a given problem and promote their thinking ability, as these learners have to choose the most appropriate solution. Lucas and Spencer (2017) have also established that experiential learning promotes learners' creative thinking skills since they are practical and motivate learners to think creatively. They further believe that the methods associated with CTS cultivate learners' thinking ability, and their learning of CTS grows naturally, encourages them to think creatively, absorb details from concrete imaginations, and helps them make informed decisions. This view was acknowledged by Gao (2015), who established that some teaching methods, such as games and puzzles, provide mental challenges that help learners to develop their CTS. This is because they allow learners to allow for trial and error so that learners can learn from their mistakes and come up with new ideas to solve problems.

The teachers in this study also believe that the methods associated with promoting CTS, such as discussion, push learners to build creative thinking ability and continually keep joining others' ideas in problem solving. In this case, learners process information to find or create solutions for the problem, rather than information from a teacher or others dictating how they should solve it. Huang et al. (2017) believe that learners work cooperatively in the methods such as discussions and acquire experience in creative thinking skills that help them to cope in dealing with complex problems.

5.2.4 Factors Leading to the Challenges that Teachers Experience in Teaching CTS

Teachers in this study claimed that they encounter challenges in practising CTS in the classroom. They claim that their efforts are dwarfed by their lack of skills and knowledge, the limited time needed for teaching CTS, the lack of resources required, and the calibre of learners they have.

Teachers' lack of knowledge and adequate skills for teaching CTS makes it difficult to develop and promote this skill in the classroom. Because they are not adequately equipped with the knowledge and skills necessary to develop CTS, they are unable to identify and use effectively the strategies that may enhance learners' development of creative thinking abilities. Subsequently, many teachers do not engage learners in CTS learning activities. Chang (2013) observes that teachers find developing CTS in the classroom difficult as they have limited knowledge and lack adequate skills. In addition, teachers, lack of skills has influenced their self-esteem in teaching these skills. For example, the study has established that teachers have low self-esteem and, for that reason, do not attempt to teach learners CTS. Suciawati (2019) had a similar discovery that teachers' poor self-confidence, low self-esteem, and a feeling of helplessness create an obstacle that hinders them from teaching CTS.

Teachers' perceptions that teaching CTS is time-consuming and their workload does not allow them to spend enough time preparing and teaching CTS is another study finding. Tangaard (2011) has also established the same view, that teachers' views on how they should spend their instructional time is one factor that influences their reluctance to engage learners in creative thinking. Beghetto and Kaufman (2012) also believe that intensive planning and designing of learning tasks for teaching CTS puts much pressure on teachers, making them reluctant to teach these skills.

How teachers view the availability of resources appears to be also influential on how they teach CTS. In this study, teachers claimed that their schools lack the resources to teach learners to become creative thinkers. Eragamreddy (2013) and Saadi (2016) also observed that the schools' lack of facilities hinders teaching that makes learners think creatively and independently. Thus, teachers claim that learners could be too slow to think without these resources, resulting in slow and tedious teaching.

The teachers in this study also believe that their learners are not exposed to situations that can make them use their imaginations and think creatively. That being the case, teachers encounter difficulties influencing learners to think creatively. This view is aligned with Chang (2013), who established that teachers find it challenging to develop CTS in the classroom of learners with limited knowledge and exposure.

5.2.4 How Teachers are Supported to Teach CTS

This study has established that teachers are not happy with the way they are supported to teach CTS. They pointed out that the prevailing situation in their schools limits the support they are supposed to receive so that they can successfully teach CTS. Teachers also complained that even the management in their school is not taking any responsibility to guide them on how they can teach and develop CTS in their classrooms. The view of these teachers is that in their schools, nobody is in control to ensure that teachers teach and engage learners in developing creative thinking skills. As a result, this skill is not taught at all in schools. Where it is taught, it is done in a limited manner. Adams (2019) has the same view that where professionals do not sufficiently support teachers, they are rarely motivated to develop the required skills on their own. The teachers also complained about the non-engagement in the professional development activities relating to the teaching of CTS. Their view is that the Ministry of Education and Training also appear not to be concerned about the success of this strategy in the classroom. Thus, The Ministry has not provided training workshops to empower them.

5.2.5 Teachers' General Views about Teaching CTS in Lesotho Classrooms

Although teachers do not generally teach CTS in their classrooms, they value this skill and indicate that the skill that can be implementable in the classrooms if the conditions in the schools could be improved. They first pointed out that these conditions could be improved through the government and the schools' efforts to provide teachers with appropriate training workshops, teaching resources, resource teachers, facilitators, and a conducive learning environment. They believe regular training workshops can equip them with the necessary skills to teach CTS. This view that teachers need training workshops that will develop their understanding of creative thinking skills through practical training has been emphasized by Hosseini and Watt (2010) and Patston et al. (2018). Teachers in this study also believed that both the government and the schools should commit to teaching this strategy and provide classrooms with appropriate

teaching resources. Teachers also identified engaging experts to support teachers in improving their CTS teaching skills as imperative. They believed that through the guidance of this human resource, they were likely to acquire the knowledge and skills required. These teachers emphasized that the Ministry of Education's engagement in providing schools with resource teachers with expertise in teaching creative thinking skills should be compulsory and a long-term commitment. Cherkasov and Smigel (2016) also advocate for the engagement of experts to develop creative thinking skills.

Teachers in this study also believe that the school environment should be conducive to CTS learning. Such an environment should encourage learners' intrinsic motivation and incentivize them to explore ideas and possibilities without being judged or ridiculed. They further emphasized that a conducive learning environment in their schools should also allow learners to experience different situations that will allow them to think creatively.

Even though the majority of participants believed that improving classroom conditions in Lesotho schools could lead to the successful teaching of CTS, a few teachers had different opinions. These teachers have argued that, despite its benefits, they find teaching CTS time-consuming, as it requires much time for preparation. They believe that the Ministry of Education and Training should be highly committed to playing a significant role in influencing the success of this strategy. Thus, the Ministry should first treat this as a pilot project in a few schools to gauge the material and equipment needs, training teachers. Further, the Ministry should be prepared to meet the schools halfway with the expenses related to teaching CTS.

5.3 CONCLUSIONS

This study concludes that from their interpretations, teachers understand what CTS means. They are aware of the benefits of developing learners' CTS. Such benefits include developing academic and personal growth and intelligence and finding innovative ways to solve complex problems in the community and workplace. Many teachers do not teach this skill in their classrooms. Those who do, employ different methods that can engage learners actively. These methods include discovery, experiential learning, word and puzzles, brainstorming, and discussions.

The study also concludes that teaching CTS in Lesotho classrooms poses many challenges. Factors leading to these challenges include limited time as perceived by teachers, teachers' lack of knowledge and skills, the calibre of learners, and lack of resources. These factors influence how teachers develop CTS in the classroom; as a result, it becomes difficult for them to develop this skill.

Another main challenge is the limited support to teach this skill that teachers receive from their schools or the Ministry of Education and Training. Despite these challenges, teachers firmly believe that if the conditions in Lesotho schools are improved, teaching CTS can be successful and effective.

5.4 LIMITATIONS OF THE STUDY

The limitation of this study was in the methodology. The study engaged teachers in only one high school located in the Leribe district. Thus, the number of teachers who participated in this study is too small to consider it to represent all teachers in the Leribe district. Even in the school where the study was held, only a few teachers participated in this study. Similarly, the views and experiences of these teachers do not represent the views and experiences of the teachers of that school. As a result, this study is relatively too small to represent all high schools in Lesotho. The study also used one method of data collection, which was a semi-structured interview. Other methods might have generated different data and the results established in this study.

5.5 RECOMMENDATIONS

Based on the conclusions drawn in this study, recommendations regarding policy practice and further research have been made. They are explained hereafter.

Recommendations for practice. The Ministry of Education and Training and other stakeholders in education should find ways of supervising and monitoring the implementation of CTS in Lesotho classrooms. The Ministry should also offer training programmes that target to develop school management and teachers' pedagogical skills relating to CTS. The curriculum designers should develop guidelines for teachers that incorporate effective strategies and methods to develop CTS in Lesotho classrooms in the curriculum. The inspectors in the Ministry should monitor and observe that the teaching and promotion of CTS is achieved in Lesotho classrooms.

Recommendation for policy. There should be a detailed policy document by the Ministry of Education and Training in Lesotho that explains how the teaching of creative thinking skills should be integrated into different subjects across the curriculum. The policy should further emphasize the effectiveness of teaching this skill in the classroom to enhance learner thinking ability.

Recommendation for further research. The study used only one method to collect data: a semi-structured interview. This method involved a small number of teachers in one school, which could not be viewed as a representative of teachers in Lesotho schools. A large number of teachers from different schools could give a broader picture of how the teaching of creative thinking skills is engaged in classrooms.

5.6 **SUMMARY**

This study investigated teachers' experiences of teaching creative thinking skills in Lesotho classrooms. This chapter discussed the results that were presented in Chapter 4. The conclusions were made based on these discussions. In this chapter, the limitations of this study have been presented, and the recommendations for policy, practice and further research have been presented.

REFERENCES

- Abawi, K. (2017). *Data* collection methods(questionnaire & interview). Paper presented at: Geneva Workshop 2017. *Training Course in Sexual and Reproductive Health Research*; 2017 Nov 28; Geneva.
- Abraham, A. (2016). Gender and creativity: An overview of psychological and neuroscientific literature. *Brain Imaging and Behavior*.10(2): 609-18. Doi: 10.1007/s11682-015-9410-8. PMID: 26051636.
- Adams, B. L. (2019). The far reaching impact of transformative curriculum: a narrative critical ethnographioc case study. *Journal of Curriculum Studies Research*, 1(1), 17-32, https://curriculumstudies.org/index.php/CS/article/view/8
- Akgunduz, D., Aydeniz, M., Cakmakci, G. Cavas, B. Corlu, N. S., Oner, T., & Ozdemir, S. (2015). A report on STEM education in Turkey: A provisional agenda or a necessity? [In Turkish]. Istanbul, Turkey: Istanbul Aydin University. Retrieved from http://hstem.hacetepe.edu.tr/STEM Rapor.pdf.
- Al-Atoom, A., AL-Jarah, A., & Beshara, M. (2011). *Developing Thinking Skills: Models and Practical Applications*. Ammah, Dar Al-Masirah (In Arabic).
- Alencar, E.M. & Fleith, D.S. (2016). Relationships between motivation, cognitive styles and perception of teaching practices for creativity. *Electronic Journal of Research in Educational Psychology*, 10(3), 1011-1034.
- Alharahsheh, H. H., & Pius, A. (2020). A review of key paradigms: positivism vs interpretivism. Global Academic Journal of Humanities and Social Science, 2(3), 39-43
- Al-Zoubei, A. (2014). *Creative Self-efficacy in Gifted Students and their Teachers in Jordan*. The Beehive, in press (In Arabic).
- Amabile, T. and M. Pratt (2016). The dynamic componential model of ccreativity and innovation in organizations. *Making Progress, Making Meaning*. .36. DO-10.1016/j.riob.2016.10.001
- Andiyana, M.A., Maya, R., & Hidayat, W. (2018). Analysis Kemampuan Berpikr Kreatif Matematis SiswaMts Pada Materi Lingkaran. Aksioma: *Jurnal Program Studi Pendidikan Matematika*, 1(3), 239-248 https://doi.org.10.24127/ajpm.v7il.1316
- Anney, V. N. (2014). Ensuring the quality of the findings of qualitative research: looking at trustworthiness criteria. *Journal of Emerging Trends in Educational Research and Policy Studies*, 5(2):272-281. Available at http://repository.udsm.ac.tz.8080/xmlui/handle/123456789/256[Accessed on 13 September 2014]
- Apino, E. & H. Retnawati, (2017). Developing instructional design to improve mathematical higher order thinking skills of students. *Journal of Physics: Conference Series*. 812, 1-7. doi:10.1088/17426596/1/0/1001.

- Applied Educational Systems. (2019). What are the 4Cs of 21st Century Skills? https://www.aeseducation.com/career-readiness/what-are-the-4Cs-of-the-21st-century-skills
- Arifin, Z. (2017). Mengembangkan instrument pengukur critical thinking skills siswa pada pembelajaran matematika abad 21, *Jurnal THEOREMS* (The Original Research of Mathematics), 1(2), 92-100.
- Arikunto, (2010). *Prosedur Penelitian*: Suatu Pendekatan Praktek. Jakarta: Rineka Cipta+ripin, I. M. K., Sugandi, I. H., Mu'minah and A. Mulyan, (2020). Pelatihan Pembelajardn Biologi Abada 21. Bernas: *Jurnal Penglabdian Kepada Masyarakat*, 1(3), 150-158.
- Arvyati, Ibrahim M, Irawan, A. (2014). effectively of peer tutoring learning to increase mathematical creative thinking ability of class xi IpaSman 3 Kendari 2014. IJER, 3(1): 613-628
- Ary, D., Jacobs, L. C., Sorensen, C. (2010). *Introduction to Research in Education* (8th ed.)In Thompson Wadsworth (ed.): Cengage Learning products, presented in Canada by Nelson Education, Ltd.
- Azhari, A., & Somakin, S. (2014). Peningkatan Kemampaun Berpikir Kreatif Matematik Siswa Melalui Pentekatan Konstruktivisme Di Kelas Vii Sekolah Menengah Pertama (Smp) Negeri 2 Banyuasim 111. *Jurnal Pendidikan Matematika*, 8(1), 1-2 https://doi.org/10.22342/jpm.8.1.99.1-12
- Babbie, E. (2010). *The Practice of Social Research*(12th Ed.). Belmont, CA: Wadsworth Cengage Learning.
- Beghetto, R.A., & Kaufman, J.C. (Eds). (2010). *Nurturing Creativity in the Classroom*. Cambridge University Press.https://doi.org/10.17/CB09780511781629
- Bereiter, C. (1994). Constructivism, socioculturalism, and popper's world 3. *Educational Researcher*, (23), No. 7, pp. 21-23. https://doi.org.3102/0013189X023007021
- Berg, B.L., & Lune, H. (2012). *Qualitative Research Methods for Social Sciences* (7th Ed). Pearson International.
- Besancon, M. & Lubart, T. (2008). Individual differences in the development of creative competencies in school children. *Learning Individual Differences*, 18, 381-389.
- Bolderston, A. (2012). Conducting a research interview. *Journal of Medical Imaging and Radiation Sciences*, 43(1), 66-76.
- Bowling, A. (2014). *Research Methods in Health: Investigating health and health services*. McGraw-Hill Education (UK)
- Braun, V., & Clarke, V. (2012). Thematicanalysis. In H. Cooper (Ed.) *APA Handbook of Research Methods in Psychology*. American Psychology
- CAP, (2009). *Education for International and Social Development*. Ministry of Education Government of Lesotho. Maseru: Morija

- Carson, S. (2010). Latent inhibitation and creativity. In R.E. Lubow & I. Weiner (Eds), *Latent Inhibition: Cognition, Neuroscience and Publications to Schizopherenia* (pp. 183-198). https://doi.org/10.1017/CB09780511730184.010
- Center for Curriculum Redesign. (2015). Skills for the 21stCentury: What should students learn? https://curriculumredisn.or/wp-content/uploads/CCR-skills-FINAL-June2015.pdf.
- Cintia, N. I. Kristin, F., & Anugraheni, I. (2018). Penerapan Model Pembelajaran discovery learning (Untuk Meningkatkan Kemampuan Berpikir Kreatif Dan Hasil Belajar Siswa). *Perspektif Ilmu Pedidikan*, 32(1), 30
- Chang, Y. S. (2013). Student technological creativity using online problem solving activities. International Journal of Technology and Design Education, 23, 803-816. https://doi.org/10.1007/s10798-012-9217-5
- Cheng, V.M.Y. (2010). Teaching creative thinking in regular science lessons: Potentials and obstacles of three different approaches in an asian context. *Asia-Pacific Forum on Science Learning and Teaching*, (11)1, 1-7
- Cheng-Shih Lin (2016). Effects of web-based creative thinking teaching on students' creativity and learning outcome Eurasia *Journal of Mathematics, Science & Technology Education*, 2016, 12(6), 1675-1684.
- Cherkasov, A.A., & Smigel, M. (2016). Public education in the Russian empire during the last third of the xix century: Parish Schools. *European Journal of Contemporary Education*, 18(4), 418-429
- Cheung, H. P. R., & Leung, C. H. (2013). Preschool teachers' beliefs of creative pedagogy: Important for fostering creativity. *Creativity Research Journal*, 25, 397-407.
- Chiumento, A., Rahman, A., Machin, L., & Frith, L. (2017). *Mediated research encounters*: methodological considerations in cross-language qualitative interviews. 18(6). https://doi.org/10.1177/1468794117730121
- Clarke, V., & Braun, V. (2017). Thematic analysis. *The Journal of Positive Psychology*, 12, 297-298. https://doi.org/10.1080/17439760.2016.1262613
- Cohen, L., Manion, L., & Morrison, K. (2011). *Research Methods in Education*. (7th ed.). London: Routledge.
- Colangelo, N., & Davis, G. (2011). *The Reference in Gifted Education*. Translated by Saleh Abo-Gado & Mahmoud Abu-Gado, Rijadh, Obeikan Publishing.
- Connelly, L.M. (2016). Trustworthiness in qualitative research. *Medsurg Nursing*. 2016 Nov; 25(6): 435-6. PMID: 30304614.
- Cope, D. (2015). Case study research methodology in nursing research. *Oncology Nursing Forum*42(6), 681-882. Doi:10.1188/15.ONF.681-682.
- Corazza, G. E. (2016). Potential originality and effectiveness: The dynamic definition of creativity. *Creativity Research Journal*, 28(3), 258-267.https://doi.org/10.1080/10400419.2016.1195627

- Corona, V. (2017). An ethnographic approach to the study of linguistic varieties used by young Latin Americans in Barcelona. In E. Moore & M. Dooly (Eds). *Qualitative Approaches to Research on Plurilingual Education*170-188. Research-publishing net. https://doi.org/10.14705/rpnet.2017.emmd2016.627
- Craft, A. (2011). *Creativity and Education Futures: Learning in a Digital Age*. Stoke-on-Trent: Trentham Books.
- Craig, T.S. (2012). Experiences of non-English students in writing for learning in mathematical problem-solving in Proceedings of ICME12-12TH. *International Congress on Mathematical Education*, Seoul, (forthcoming)
- Cresswell, J.W. (2012). Educational Research: Planning, conducting, and evaluating quantitative and qualitative research (4th ed.). Boston, MA: Pearson
- Cresswell, J.W. (2013). *Qualitative Inquiry & Research Design*: Choosing among Five Approaches (3rd ed.) Thousand Oaks, CA: SAGE Publications, Inc.
- Cresswell, J.W. (2014). *Research Design: Qualitative, Quantitative and Mixed Methods Approaches* (4th ed.).Thousand Oaks, CA: SAGE
- Cronin, C. (2014). Using case study as rigorous form of inquiry. *Nurse Research*, 21(5): 19-27. Doi: 10.7748/nr.21.5.19.e1240
- Daud, A. M., Omar, J., Turiman, P. & Osman, K. (2012). Creativity in science education. *Procedia-social and. Behavioral Sciences*, *59*, 59467. https://doi.org/10.1016/j.sbspro.2012.09.302
- David, M. & Sutton, C.D. (2004). Social research the basics. In *Social Research* London: SAGEPublications. ISBN 0761973672, 9780761973676.
- Directions Evidence and Policy Research Group, LLP. (2014). Ontario Regulation 274-Final report.http://www.edu.gov.on.ca/eng/policyfunding/memos/nov2014/REG274EN.pdf
- Doody, O., & Noonan, M. (2013). Preparing and Conducting Interviews to Collect Data. *Nurse researcher*, 20, 28-32.https://doi.org/10.7748/nr2013.05.20.5.28.e327
- Elster, J. (2007). Explaining Social Behavior: More Nuts and Bolts for the Social Sciences.

 CambridgeUniversityPress. http://www.faculty.rsu.edu/fe/well/Theorists/Weber/Whome 2.htm
- Ennis, R.H. (2011). The nature of critical thinking: an outline of critical: Thinking dispositions and abilities. The Revised Version of a Presentation at the Sixth International Conference on Thinking at MIT Cambridge.
- Eragamreddy, N. (2013). Teaching Creative Thinking Skills. IJ-ELTS 1(2), 124-145.
- Ershadi, M. & Winner, E. (2020). Children's creativity. In M. A. Runco & S.R. Pritzker (Eds). Encyclopedia of Creativity. 1, 144-147)
- Eshet, Y. & Margaliot, A. (2022). *Does Creative Thinking Contribute to the Integrity of Education students?* DOI:10.3389/FPSYG.2022.925195

- Facione, P.A. (2013). *Critical Thinking: What it is and why it counts*. Millbrae, CA: Measured Reason and the California Academic Press.
- Flick, U. (2011). *Introducing Research Methodology: A Beginner's Guide to Doing a Research Project*. Los Angeles: SAGEPublications.
- Flick, U. (2014). *An Introduction to Qualitative Research.5th Ed*, SAGE Publications, London.
- Gafour, A. S. and Gafour, W. A. (2020). *Creativity Thinking Skills- A Review article*. https://researchgate.net/publication/349003763
- Gao, (2015). Correlation analysis of influence of mathematical games on creativity. *Course Education Research*, 12, 257-258
- Gardner, H. (2011). Creating Minds: An anatomy of creativity seen through the lives of Freud, Einstein, Picasso, Stravinsky, Eliot, Graham, and Gandhi. Basic Books
- Geronik, L. & Kimhi, Yael. (2019). *Creativity Promotion in an Excellence Program for Preservice Teacher Candidates*. 71(5). https://doi.org/10.1177/0022487119873863
- Gray, D. E. (2014). Doing Research in the Real World. London: SAGE
- Gregory, E. Hardiman, M., Yarmolinskaya, J., Rinne, L. & Limb, C. (2013). Building creative thinking in the classroom: From research to practice. *International Journal of Educational Research*, 43-50. Doi:10.1016/j.ijer.2013.06.003
- Griffin, P., McGaw, B., & Care, E. (2012). The changing role of education and schools.In P. Griffin, B. McGaw, & E. Care (Eds.) *Assessment and Teaching of 21st Century Skills*.1-16. http://dx.doi.org/10.1007/978-94-007-2324-5-2
- Grohman, M.G. & Szmidt, K.J. (2013). Teaching for creativity: How to shape creative attitudes in teachers and students. In: M.B. Gregorson, H.T. Snyder & J.C. Kaufman (Eds), *Teaching Creatively and Teaching Creativity*: Springer Science Media.
- Gunduz, N. and Hursen, C. (2015). Constructivism in teaching and learning:Content analysis evaluation, procedia-social and behavioral sciences. *The Proceedings of 6th World Conference on Educational Sciences*, 191; 526-533
- Hammersley, M. (2010). What is Qualitative Research? London. New Delhi.New York: Sydney Bloomsbury.
- Harris, D. N., & Sass, T. R. (2011). Teacher Training, Teacher Quality and Student Achievement. *Journal of Public Economics*, Elsevier. 95(7-8), 798-812.
- Harry, B., & Lipsky, M. (2014). Qualitative research on special education teacher preparation. In
 M. McCray, T. Brownell, & B. Lignugaris/ Kraft (Eds), Handbook of research on special education teacher preparation. 445-460
- Hennink, M. M., Hutter, I., & Bailey, A. (2011). *Qualitative Research Methods*. SAGE Publications, London, Los Angeles, New Delhi, Singapore, Weshington DC.

- Herrmann, E. (2015). *The 4 C's of 21stCentury Learning for ELLs: Critical Thinking*. Retrieved from https://exclusive.multibriefs.com/content/the4-cs-of-21st-century-learning-for-ells-critical-thinking/education.
- Hobbs, R., Ebrahimi, A., Cabral, N., Yoon., J. & Al-Humaidan, R. (2010). *Combatting Middle East Stereotypes through Media Literacy Education in Elementary School*. Unpublished manuscript, Media Education Lab, Temple University
- Hosseini, A., & Watt, A. (2010). The effect of a teacher professional development in facilitating Students' Creativity. *Educational Research*, 5(8), 432-438.
- Huang, M. Y., Tu., H. Y. Wang., W. Y., Chen, J. F., Yu, Y. T. & Chou, C.C. (2017). Effects of cooperative learning and concept mapping intervention on critical thinking and basketball skills in elementary school. *Thinking Skills and Creativity*, 23, 207-216. https://doi.org/10.1016/j.tsc.2017.01.002
- Hursen, C., Kaplan, A. and Ozdal, H. (2014). Assessment of creative thinking studies in terms of content analysis. *Procedia-Social. Behav. Sci.* 143 1177.
- Husaman, H. (2015). Thinking skills for environmental sustainability perspective of new students of biology education department through blended project based learning model. *Journal Pendidikan IPA Indonesia*, 4(2), 110-119. https://doi.org/10.15294/jpii.v4:2.3878
- Hyett, N. Kenny, A., & Dickson-Swift, V. (2014). Methodology or method? A critical review of qualitative case study reports. *International Journal of Qualitative Studies on Health and Well-being*, 9, 23606.
- Jacobs, L. & T. Tlali (2015). Educational in the Mountain Kingdom of Lesotho. In Wolhuter, Cc, Jacobs, L. & Steyn, HC. (Eds). *Thinking about Education Systems*, 2nd Edition. (pp.78-109). Potchefstroom: Keurkopie
- Jeng, y., Hsu, S., Xie, J., & Lin, R. (2010). *Notice of Retraction and the Influence of Creative Thinking Teaching on Learning Effectiveness*. Retrieved December 18, 2014, http://ieeexplore.ieee.org/xpl/articleDetails.jsp?arnumber=5553294
- Johnson, B., & Christensen, L. B. (2008). *Educational Research: Quantitative, Qualitative and Mixed Approaches*. Los Angeles: SAGE Publications
- Juliantine, T. (2019). Pengembangan kreativitas siswa melalui mmplementasi model pembelajaran inkuiri Dalam Pendidikan Jasmani https://doi.org/10.1017/CB09781107415324.004.
- Karpov, Y.V. (2014). *Vygotsky for Educators*. Cambridge: Cambridge University Press. https://doi.org/10.1017/CB09781107588318.
- Kassema, J.J. (2016). Knowledge management implementation impacts In Tanzania's public sector: Case study Ministry of Finance and Planning. *Social Science Research Network*, 5-10. SSRN Abstract ID. 2790205
- Kaufman, J.C., Beghetto, R. A., & Dilley, A. (2016). Understanding creativity in the schools. In psychosocial skills and schools systems in the 21st century. *International Publishing*

- Kettler, T., Lamb, K. T., Wilerson, A., & Mullet, D.R. (2018). Teacher's perceptions of creativity in the classroom. *Creativity Research Journal*, 30(2), 164-171
- Khoiri, A. & Sunarno, W. (2019). *How is students' creative thinking skills?* An ethnoscience learning implementation 08 (October), 153-163. https://doi.org/10.24042/jifalbiruni 0i0.4559.
- Kivunja, C. (2015). Exploring the pedagogical meaning and implications of the 4Cs "Super Skills" for 21st century through Bruner's SE lenses of knowledge construction to improve pedagogies of the new learning paradigm. *Journal of Creative Education*, 6,224-239.DOI:10.4236/ce.2015.62021
- Kuhn, D. & Moore, W. (2015). *Argument as core curriculum learning*: research and practice. *Advance online publication*.doi:10.1080/23735082.2015.994254.
- Kumar, R. (2011). *Research Methodology: A Step-by-Step Guide for Beginners*. 3rd Edition. SAGE, London. New Delhi.
- Laius, A. & Rannikmae, M. (2011). Impact on student change in scientific creativity and socioscientific reasoning skills from teacher collaboration and gains from professional inservice. *Journal of Baltic Science Education* 10(2): 127-137
- Lam, R. (2015). Language Assessment Training in Hong Kong: Implications for language assessment literacy. Language Testing, 32(2), 169-197. http://dx.doi.org/10.1177/0265532214554321
- Lambert, S. D. & Loiselle, C. G. (2008). Combining individual interviews and focus groups to enhance data richness: *Journal of Advanced Nursing*, 62(2), 228-237
- Larraz-Rabos, N. (2021). *Development of Creative Thinking Skills in the Teaching-Learning Process.* DOI: 10.5772/intechopen.97780
- Li, C. (2016). Effects of web-based creative thinking teaching on students' creativity and learning outcome Eurasia. *Journal of Mathematics, Science & Technology Education*.
- Louw, A., Lawrence, H. & Motto, J. (2014). 'Mammographer Personality Raits- Elements of the Optimal Mammogram Experience', *Health SA Gesondheid* 19(1), Art. #803, 7 pages. http://dx.doi.org/10.4102/hsag.v19i1.803
- Lubart, T., & Batton, D. (2017). Enhancing Creativity. In T.S. Yamin, K. W. McCluskey, T. Lubart, D. Ambrose, K. Mccluskey, & S. Link (Eds.), Innovation Education (pp. 119-128). Sweden: ICIE the International Centre for Innovation in Education.
- Lucas, B., & E. Spencer (2017). *Teaching Creative Thinking: Developing Learners Who Generate Ideas and Can Think Critically.* Wales: Crown House Publishing Ltd, ISBN 978-1-78583-267-3.
- Lucas, B., & Venckute, M. (2020). Creativity- ATransversal Skill for Lifelong Learning. An Overview of Existing Concept and Practices, Kampylis, P. and Cachia, R. editor(s), EUR 30405 EN, Publications Office of the European Union, Luxembourg, 2020, ISBN 978-92-76-23323-7, doi:10.2760/557196, JRC 121862.

- Lucas, B., G. Claxton, & E. Spencer. (2016). Progression in student creativity in school: First steps towards new forms of assessments, *OECD Education Working Papers*, NO. 86, OECD Publishing, Paris https://dx.doi.org/10.1787/5k4dp59mvdwk-en.
- Maksum, A. Wayan Widiana, I., & Marini, A. (2021). Path analysis of self-regulation, social skills, critical thinking and problem-solving ability on social studies learning outcomes. *International Journal of Instruction*, 14(3), 613-628.
- Magald, D. & Berler, M. (2020). Semi-structured Interviews. In V.Zeigler-Hill, Shackelford T.K.(eds) *Encyclopedia of Personality and Individual Differences*. Springer, Cham. https://doi.org/10.1007/978-3-319-24612-3-857 Accessed on July 5th, 2021.
- Margaliot, A. N. Magid (2020). The role of creative thinking in the process of forming teachers' professional identity. *Creative Education* 11(7). DOI: 10.4236/ce.2020. 117074 PDF HTML XML.
- Maxwell, J.A. (2012). *Qualitative Research Design: An Interactive Approach*. SAGE Publications, Inc., Thousand Oaks, CA.
- Merriam, S. B. (2009). *Qualitative Research: A Guide to Design and Implementation*. San Francisco, CA: Jossey-Bass.
- Merriam, S. B., & Tisdell, E. J. (2015). *Qualitative Research: A Guide to Design and Implementation*. San Frascisco, CA: John Wiley & Sons.
- Mishra and Kereluik (2013). What 21stCentury Learning? A Review and a Synthesis. In Proceedings of SITE 1011-Society for information Technology & Teacher Education International conference, 3301-12 Nashville: Association for the Advancement of computing in Education.
- MoET (2009). *Curriculum and Assessment Policy: Education for Individual and Social Development.* http://plainipolis.iiep.unesco.org/sites/planipolis/files.
- Murphy, M. (2014). What are the Benefits and Drawbacks of Case Study Research? [Website]. http://socialtheoryapplied.com/2014/05/24benefitsdrawbacks-case-study-research/ [2016-12-02].
- Mvududu, N.H. & Thiel-Burgess, J. (2012). Constructivism in practice: The Case for English language learners. *International Journal of Education*, 4(3), p108-p118. http://dx.doi.org/10.5296/ije.v4i3.2223
- National Education Association. (2015). *Preparing 21st Century Students for a global society*: An educator guide to the "Four Cs". Retrieved from https://www.nea.org/org/assets/docs/A-Guide-to-Four-Cs
- Naylor, S. & Keogh, B. (1999). Constructivism in classroom: Theory into practice. *Journal of Science Teacher Education* 10 (2), 93-106
- Nuzliah, N. (2015). Kontribusi motivasi belajar, kreativitas terhadap problem solving (pemecaham masalah) siswa dalam belajar serta implikasi terhadap bimbingan dan koseling di SMPN 29 Padag. *Jurnal Edukasi*, 1(2), 157-174.

- OECD (2019). Framework for the Assessment of Creative Thinking in PISA 2021: Third Draft. OECD Publishing.
- Palmer, T. (2015). *15 Characteristics of a 21stCentury Teacher*. Retrieved from https://www.edutopia.org/discussions/15-characteristics-21st-century-teacher.
- Patston, T.J., D. H., Cropley, R.L., Marrone & J.C. Kaufman (2018). Teacher implicit beliefs of creativity: Is there an arts bias? *Teaching and Teacher Education*, 75 (2018), pp. 366-374, 10.1016/j.tate.2018.08.001
- Pesout, O., & Nietfeld, J. (2021). The impact of cooperation and competition on metacognitive monitoring in classroom context. *Journal of Experimental Education*, 89(2), 237-258. https://doi.org/10.1080/00220973.2020.1751577.
- Piaget, J. (1968). Part I: Cognitive development in children: Piaget development and learning. Journal Research in Science Teaching, 2(3), 176-186. Doi: 10.1002ltea.3660020306
- Plucker, .A.J., M.A., Runcoab, M.A., & Simonsen (2020). *Enhancement of creativity: The Curated Reference Collection in Neuroscience and Biobehavioral Psychology* (3rd ed.). https://doi.org/10.1016/B978-0-12
- Polit, D. F., & Beck, C.T. (2014). Essentials of nursing research: appraising evidence for nursing Practice. (8th ed.). Philadelphia, Wolters Kluwer Health/Lippincott Williams & Wilkins.
- Puccio, G., & Cabra, J.F. (2010). Organizational creativity: a systems approach. In. J.C. Kaufman, & R. J. (Eds), *The Cambridge Handbook of Creativity*. 145-173. New York: Cambridge UNIVERSITY Press.
- Rakotsoane, F. C. L. (2012). Writing a scholarly Research proposal: *A Step-by-Step Guide for Young Researchers*. Lesotho: Morija Printing Works.
- Rasanen, M. & Nyce, J. M. (2013). The raw is cooked: Data in intelligence practice; Science, Technology and Human Values 38(5), 655-677. https://doi.org/10.1177/0162243913480Research. Oncology Nursing 42(6), 681-882.
- Raselimo, M. & Mahao, M. (2015). The Lesotho Curriculum and Assessment Policy: Opportunities and Threats. *South African Journal of Education*, *35*(1).
- Reunamo, J. (2014). Children's creativity in day care. *Early Child Development and Care*, 184(4). 617-632. Doi: 10.1080/03004430.2013.806495.
- Richardson, A. J. (2012). Paradigms, Theory and management accounting practice: A comment on Parker (forthcoming). Qualitative management accounting research: Assessing deliverables and relevance. *Critical Perspectives on Accounting*, 23(1), 83-88 https://dx.doi.org/10.1016/j.cpa.2011.05.003
- Ritter, S. M., & Mostert, N. (2016). Enhancement of creative thinking skills using cognitive-based creativity training. *Journal of Cognitive Enhancement, 243-253, doi:* 10.1007/s41465-016-0002-3
- Robinson, K. (2011). *Out of Our Minds: Learning to be Creative*. West Sussex, UK: Capstone Publishing.

- Robinson, K. (2015). Creative schools. INAllen Lane. *The Grassroots revolution that's transforming education*. Great Britain:Viking
- Sarwinda, W. (2013). Pemberdayaan Keterampilan berfikir kreatif siswa melalui strategi reciprocal teaching pada pembelajaran biologi SMA [Empowerment of student's creative thinking skills through reciprocal teaching strategies in high school biology learning]. Inc. Kubikazari & H. Rahmawati (Eds), *National Seminar X Biology Education* (pp.14-139).
- Saumure, K. & Given, L. (2008). Convenience sample.In L. M. Given (ed.). The SAGE *Encyclopedia* of Qualitative Research Methods, pp.125-126, SAGE, Thousand Oaks, CA.
- Saunders, M., Lewis, P. & Thornhill, A. (2012). *Research Methods for Business Students*.(6th ed.) Pearson Education Limited.
- Schleicher, A. (2012). *Preparing Teachers and Developing School Leaders for the 21st Century*: Lessons from around the World, OECD Publishing. http://dx doi.org/10.1787/9789264xxxxx-en
- Scott, C.L. (2015). The futures of learning: What Kind of Pedagogies for the 21st century? Education Research and Foresight.https://doi.org/10.1016/jpse.2015.08.005
- Scott, L.S. (2015). The futures of learning 2: What kind of learning for 21st century?(ERF Working Paper No.14). Paris: UNESCO Education Research and Foresight. https://unesdoc.org/ark:/48223/pf0000242996
- Selkrig, M. & Keamy, K. (2017). Creative pedagogy: Acase for teachers' creative learning being at the centre. *Teaching Education* 28(3), 1-16. Doi:10.1080/10476210.2017.1296829
- Senel, M. (2019). Development of creative thinking skills of students through journal writing. *International Journal of Progressive Education*, 15(5).
- Shaheen, R.K. (2010). *Creativity and education. Creative Education.* 1,166-169.https://10.4236/ce.2010.13026
- Shannon-Baker, P. (2023) Qualitative, Multimethod, and Mixed Methods Research. in *International Encyclopedia of Education*(4thed.)
- Sharma, B., & Ali, Arif (2015). Spiritual intelligence, self-esteem and mental status among the school going adolescents. *Indian Journal of Positive Psychology*, 6 (3), 233-237
- Silverman, D. (2016). *Qualitative Research*: Issues of *Theory, Method and Practice*. (3rd Ed). London: SAGEPublications Ltd.
- Songkhram, N. (2015). *Creating Innovation: Changed Learners to Innovators.* Chulalongkorn University Press.
- Stahl, N. A., & King, J. R. (2020). Expanding approaches for research: understanding and using trustworthiness in qualitative research. *Journal of Development Education* 44(1), 26-28, 2020.https://www.researchgate.net/publication/346425936
- Sternberg, R. J., & Kaufman, S. B. (Eds). (2011). *The Cambridge Handbook of Intelligence*. Cambridge University Press. https://doi.org/10.1017/CB09780511977244

- Sternberg, R.J. (2001). Teaching for creativty. In Beghetto & J.C. Kaufman (Eds), *Nurturing creativity in the classroom*. 394-414. London: Cambridge University.
- Sternberg, R.J. (2007). Teaching psychology students that creativity is a decision. *The General Psychologist*. 36(1), 8-11
- Suciawati, V. (2019). Pengaruh self- efficacy terhadap kemampuan berpikir kreatif matematik siswa. *Journal Didactical Mathematics*, 2(1) 17-22, https://doi.org110.35697/jrbi,v2i2.76
- Swayer, R.K. (2015). A call to action: *The Challenges of Creative Teaching and Learning*. University of North Carolina and Chapel Hill.
- Taherdoost, H. (2016). Sampling methods in research methodology: How to choose a sampling technique for research. *International Journal of Academic Research in Management*, 5(3), 28-36
- Tangaard, L. (2011). Stories about creative teaching and productive learning. *European Journal of Teacher Education*. 34(2), 217-230. http://dx.doi.org/10.1080/02619768.2011.558078
- Tangaard, L. (2013). The sociomateriality of creativity in everyday life. *Culture and Psychology*. 19(1) 19, 20-32. https://doi.org/10.1177/1354067X12464987
- Tanujaya, B. (2016). Development of an instrument to measure higher order thinking skills in senior high school mathematics instruction. *Journal of Education and Practice*. 7(21), 144-148.
- Tashakkori, A., Johnson, R.B., & Teddlie, C. (2021). Foundations of mixed methods research: Integrating quantitative and qualitative approaches. In *The Social And Behavioral Sciences* (2nd ed). SAGE Publications.
- Taylor, B., Killick, C. & McGlade, A. (2015). *Understanding and Using Research in Social Work*. SAGE, Los Angeles, CA.
- Tendrita, M., Mahanal, S. & Zubaidah, S. (2016). Pemberdayaan Keerampilan berpikir kreatif melalui model remap think pair share. In *Proceeding Biology Education Conference* 13, 285-291.
- The Partnership for 21st Century skills of learning (2016). *Framework for 21stCentury Learning*. http://www.p12.Org/storage/documents/docs/p12 framework, 0816.pdf.
- Thomas, E. & Magilvy, J.K. (2011) qualitative rigor or research validity in qualitative research. *Journal for Specialists in Pediatric Nursing*, 16, 151. https://doi.org/10.111/j.1744-6155.2011.00283.x
- Thomson, S.B. (2011). *Qualitative Research*: Validity. JOAAG, 6(1),77-82.
- Tomljenovic, Z. & Tatalovic Vorkapic, S. (2020). Constructivism in visual arts classes. *Center for Educational Policy Studies Journal*. 10(4):13-32.https://doi.org/10.26529/cepsj.913
- Tripathi, V. (2011). The relevance of Tagore's concept of education in contemporary times. *International Referred Research Journal, 1* (17),107-108

- Trochim, W.M.K. and Donnelly, J.P. (2008). *The Research Methods Knowledge Base*. In Mason (ed.) *Atomic Dog*. 56-65
- Ulfa, U. Prabawanto, S., & Jupri, A. (2017). Students' mathematical creative thinking through problem posing learning. *Journal of Physics:* Conference Series, 895(1), 1-7, https://doi.org/10.1088/1742-6596/895/1/012097
- UNESCO (2020). *Educational for Sustainable Development*, n.d., https://en.unesco.org/themes/education-sustainable-development.
- UNESCO. (2014). Creativity. *In Guiding Principles for Learning in the Twenty First Century*. 15-18). www.ecolint.ch/file/621/download? Token=ZhQHu7g
- Vygotsky, I. (1986). *Thought and Language*. A.Kozulin (Ed. & Trans). Cambridge, MA. The Massachusetts Institute of Technology. http://s-f-walker.org.UK/pubsebooks/pdfs/Vygotsky-Thought-and-Language-pdf
- Vygotsky, L.S. (2016). The Psychology of Art. Cambridge, Massachusetts: M.L.T Press
- Wadaani, M. (2015). Teaching for creativity as human development towards self-actualization: The essence of authentic learning and optimal growth for all students. Creative Education, 6,669-679. https://doi.org/10.4236/ce.2015.67067
- Winfree, L. (2017). *Constructivism & Behaviorism: Teaching Students with Learning Disabilities*. https://study.com/academy/lesson/constructivism-behavoirism-teaching-students-with-learning-disabilities.html
- Woolfolk, A. (2010). Behavvioral Views of learning. In A. Woolfolk (ed.). *Educational Psychology* (11th Ed.) Columbus, OH: Pearson/Allyn & Bacon.
- Yin, R.K. (2011). *Qualitative Research from Start to Finish*. New York: Guilford. The Guilford Press.
- Yusnaeni, Corebima, A.D., Susilo, H., & Zubaidah, S. (2017). Creative thinking of low academic student undergoing search solve create and share learning integrated with metacognitive strategy. *International Journal of Instruction*, 10 (2), 245-262.
- Zabelina, D. I. & Robinson, M. D. (2011). Creativity as flexible cognitive control. *Psychology of Aesthetics, creativity, and the Arts*, 4(3), 136-143.
- Zarkasyi, W. (2015). Penelitian Pendidikan Matematika. Rafika Aditama

APPENDICES

APPENDIX 1: THE LETTER TO THE PARTICIPANT

Dear Participant

I am 'Maretsepile Maureen Molete, and I am undertaking a study with Dr Julia Chere-Masopha,

in the department of Educational Foundation at the National University of Lesotho. This is a self-

sponsored study. We are humbly inviting you to participate in this study entitled: Teachers'

Experiences of Teaching Creative Thinking Skills in Lesotho Classroom.

I am undertaking this study to investigate how teachers interpret creative thinking skills in

Lesotho classroom; their practices in teaching creative thinking skills; factors leading to the

challenges that teachers experience in teaching creative thinking skills, and how these teachers

are supported. The study focused on eight teachers who teach different subjects with different

teaching experiences.

The participants were chosen under the presumption that they would cooperate with the

research and provide information that would be useful to the study. The semi-structured

interviews questions will be answered on the papers and given to the participants to answer on

the provided space.

Participation in this study is voluntary. If answering a question makes you uncomfortable, you

are not required to. Declining at any point is completely free of consequences. If you decide to

take part, you will be given a consent form to sign. Your identity in this study will be withheld to

ensure that you are not put at risk in any way. This information will be confidential.

Should you need more clarification on issues regarding to this study, you are free to contact my

supervisor Dr. Chere-Masopha at +266 57756658

Yours Sincerely

'Maretsepile Molete (Mrs.)

77

APPENDIX 2: PARTICIPANTS' STATEMENT OF CONSENT

I agree to participate in this study. The research study and the procedures that will be conducted have been explained to me and understood. My participation in this study is voluntary and I may choose to answer only the questions I feel comfortable with, and I can withdraw participation at any timewhen I feel like.

Name of Participant:	
Signature:	
Date:	

APPENDIX 3:SEMI-STRUCTURED INTERVIEW PROTOCOL

Study Title: Teachers' Experiences of Teaching Creative Thinking Skills in Lesotho Classrooms

Introduction: This information aims at obtaining the insight into how you interpret creative thinking skills in Lesotho classrooms. In this interview, you will be asked about your views on how creative thinking skills are taught in the classroom, the factors that influences how you teach this skill, how you are supported to teach this skill and the challenges that you encounter when teaching creative thinking skills. Your participation in this study is voluntary and confidential. With your permission, the interview questions will be answered on the space provided.

Section 1: Personal Background

This section collects information about your personal and educational details. Please tick the appropriate option in the brackets provided.

Biological Traits

(a)	Age:
	[] From 30-39
	[] From 40-49
	[] From 50-59
(b)	Gender:
	[] Male
	[] Female
Educat	ional Background
(a)	Highest teaching qualifications
	[] Certificate
	[] Diploma in Education

[] B. ED

[] B. ED Hons
[] Masters' Degree
[] Doctorate
Teaching Experience(numbers of years of teaching)
[] 8 to 15 years
[] 15 to 20 years
[] 20 to 25 years
[] 25 to 30 years
[] 30 to 35 years
Feaching Specialization
[] English Language and Development Studies
[] Business Education, Accounting, and Economics
[] Mathematics and Chemistry
[] Biology and Physical Science
[] Commercials
[] Agriculture and Science

APPENDIX 4: FACULTY LETTER

THE NATIONAL UNIVERSITY OF LESOTHO

Telephone: +266 22340601/3631 Fax: +266 22340000

http://www.nul.ls



P.O. Roma 180 Lesotho Africa

Postgraduate Research Studies

FACULTY OF EDUCATION

23 November 2022

Dear School Principal,

My name is 'Mathabo Julia Chere-Masopha, a Postgraduate Research Programme Coordinator and the supervisor of Ms. Maretšepile Molete (200202785) whom I would like to introduce to you. Ms. Molete is a postgraduate student who is enrolled in the Master of Education Programme in the Faculty of Education. As part of her studies, Ms. Molete is investigating 'Teachers Experiences of Teaching Creative Thinking in Lesotho classrooms.' The Faculty has approved her research proposal and she has identified your school/schools in Leribe District as a site for possible data collection for her study. On behalf of the Faculty of Education and Ms. Molete, I am making a request that your good office gives her an opportunity to collect data for her proposed study.

Should you require additional information regarding M. Molete and her study, please contact Dr. Chere-Masopha through one of the following contacts:

Mobile #: 5775 6658

Email address: juliachere@gmail.com.

Yours Faithfully,

Julia Chere-Masopha (Ph.D)

APPENDIX 5: THE LETTER FROM THE RESEARCHER TO THE PRINCIPAL

I am Maretsepile Maureen Molete, and I am doing research under the supervision of Dr. Chere-Masopha in the Educational Foundations Department at the National University of Lesotho. I therefore request the participation of your school teachers in the study entitled: Teachers' Experiences of Teaching Creative Thinking Skills in Lesotho Classrooms.

The study will help teachers to develop their learners' creative thinking skills in their subjects. It will also help the Ministry of Education and Training to implement this skill effectively in schools. Through understanding of teachers' creative thinking skills, the ministry may develop the strategies to train teachers on creative thinking skills. Therefore, I request your support to grant me permission to conduct interviews in your school.

Yours sincerely

Maretsepile Molete

(Master of Education student at NUL)

APPENDIX 6: GUIDING INTERVIEW QUESTIONS

Research Question 1: How do teachers in Lesotho interpret creative thinking skills?

- (a) What do you understand by creative thinking?
- (b) How important is it to you that learners in your subject area develop creative thinking skills?

Research Question 2: How do teachers teach creative thinking skills in their classrooms?

- (a) How do make sure that your learners develop creative thinking skills in your classroom
- (b) Which teaching methods do you use to teach creative thinking skills
- (c) Which learning activities do you normally assign the learners?
 - a. How do you support your learners to ensure that they acquire these skills?
 - b. How do the methods you use provide learners with opportunities to develop creative thinking skills?
- (d) How effective are these methods?
- (e) What influences how you teach creative thinking skills?
- (f) How do your skills and knowledge influence how you teach creative thinking skills?
- (g) How do resources influence how you teach creative thinking skills?

Research Question 3: Which are factors that lead to the challenges that teachers experience in teaching creative thinking skills?

- (a) Which are factors that influence how you teach creative thinking skills?
- (b) What are the challenges that you encounter when developing creative thinking skills in your classroom?

Research Question 4: How are teachers supported to teach creative thinking skills?

(a) Are you currently supported to teach creative thinking skills in your school?

Research Question5: What are your views about teaching creative thinking in Lesotho classrooms?

- (a) What are your general views about the teaching of creative thinking skills?
- (b) Do you think this skill is implementable? Why?

APPENDIX7: LANGUAGE EDITOR'S LETTER



National University of Lesotho
Faculty of Education
Department of Educational Foundations
P O Roma 180
Lesotho

27 July 2023

Dr. J. Chere-Masopha
The Faculty of Education
Department of Educational Foundations
National University of Lesotho
P O Roma 180
LESOTHO

Dear Madam

Re: Copy-editing of Mrs Maretsepile Molete's M Ed dissertation titled: *Teachers Experiences of Teaching Creative Thinking Skills in Lesotho Classrooms*

I have copy-edited the above captioned dissertation by Mrs. Molete's to the best of my ability. If there are any errors, omissions and other mistakes are solely the responsibility of the author.

Yours sincerely

Tankie Khalanyane (Mr)

Senior Lecturer, EDF & B Ed Honours Coordinator

APPENDIX 8: TURNITIN CERTIFICATE

-	Lesotho Classro	ooms	ing Creative Thi		
	13% SIMILARITY INDEX	10% INTERNET SOURCES	2% PUBLICATIONS	7% STUDENT PAPERS	
	RIMARY SOURCES				
	Submitte Student Paper	d to Eiffel Corp	oration		2%
Ī	2 scholar.ui	fs.ac.za			2%
	researchs Internet Source	pace.ukzn.ac.	za		1%
E	core.ac.ul	(1%
E	www.ncbi	nlm.nih.gov			<1%
6	www.scirp	.org			<1%
7	Submitted Student Paper	to University	y of KwaZulu-	Natal	<1%
8	ojs.unimal	.ac.id			<1%
9	scholarwo	rks.wm.edu			<19